

Document de travail du LEM Discussion paper LEM

2017-05 version révisée

Does practicing literacy skills improve academic performance in first-year university students? Results from a randomized experiment

Estelle BELLITY

Fabrice GILLES

fabrice.gilles@univ-lille1.fr - LEM UMR 9221

Yannick L'Horty

URL de téléchargement direct / URL of direct download: http://lem.cnrs.fr/IMG/pdf/dp2017-05.pdf















FEGASS

Les documents de travail du LEM ont pour but d'assurer une diffusion rapide et informelle des résultats des chercheurs du LEM. Leur contenu, y compris les opinions exprimées, n'engagent que les auteurs. En aucune manière le LEM ni les institutions qui le composent ne sont responsables du contenu des documents de travail du LEM. Les lecteurs intéressés sont invités à contacter directement les auteurs avec leurs critiques et leurs suggestions.

Tous les droits sont réservés. Aucune reproduction, publication ou impression sous le format d'une autre publication, impression ou en version électronique, en entier ou en partie, n'est permise sans l'autorisation écrite préalable des auteurs.

Pour toutes questions sur les droits d'auteur et les droits de copie, veuillez contacter directement les auteurs.

The goal of the LEM Discussion Paper series is to promote a quick and informal dissemination of research in progress of LEM members. Their content, including any opinions expressed, remains the sole responsibility of the authors. Neither LEM nor its partner institutions can be held responsible for the content of these LEM Discussion Papers. Interested readers are requested to contact directly the authors with criticisms and suggestions.

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorization of the authors.

For all questions related to author rights and copyrights, please contact directly the authors.

Does practicing literacy skills improve academic performance in first-year university students? Results from a randomized experiment

Estelle Bellity*, Fabrice Gilles†, Yannick L'Horty‡

Abstract

We evaluate the impact of encouraging students to practice literacy skills, as well as improvement in these skills, on academic performance in first-year university students. Several previous studies have attempted to understand drivers for academic success in university students. To our knowledge, none focus on directly analyzing the relations between such factors and students' academic performance. We used a randomized experiment based on an encouragement design with a group of first-year students in Economics and Management in two French universities. For measuring the effects of encouragement, we included an innovative pedagogical tool for practicing literacy skills via a web platform, called *Projet Voltaire*. This tool also allowed us to get a good measure of the literacy skills of the students, both at the beginning and at the end of the first term of the academic year. During the entire semester, students had the opportunity to practice literacy skills using *Projet* Voltaire. To evaluate the impact of literacy on different final grades or final exam scores, and particularly on first-year grade averages, we distinguished between two randomly selected groups of students: some were encouraged to practice literacy skills, while others were only made aware of the option. As a measure of improvement in literacy skills, we use the difference between scores on the two literacy tests. Estimating intention to treat and local average treatment effect, we show that both encouragement to practice literacy skills and an improvement in literacy test scores over the first term are positively correlated with the academic performance of first-year university students, and in particular the probability that they will complete one or both semesters of the academic year.

This study benefited from financial support from University Paris-Est Marne-La-Vallée (Paris region; France) through BQR (French Bonus for the Quality of Research) funds in 2014, awarded to study the relation between literacy skills and academic failure in university students. It also benefited from support from University Lille 1 (Northern France) through funds for Restructuring and Educational Innovations in 2013. We are grateful to participants in the 65th Annual Meeting of the French Economic Association (AFSE, 2016; France), the 25th Meeting of the Economics of Education Association (AEDE, 2016; Spain), and the annual Conference on Evaluation of Public Policies (TEPP, 2016; France). The authors have benefited from helpful comments and suggestions from these audiences. Finally, we would like to thank Laurent Sarfati for his considerable work on our database.

[†]Fabrice GILLES, Université Lille 1, FRANCE; LEM-CNRS (UMR 9221) and TEPP-CNRS (FR 3435), fabrice.gilles@univ-lille1.

^{*} Estelle BELLITY, Université Paris-Est Marne-La-Vallée, FRANCE; ERUDITE and TEPP-CNRS (FR 3435), estelle.bellity@u-pem.fr.

[‡] Yannick L'HORTY, Université Paris-Est, FRANCE; ERUDITE and TEPP-CNRS (FR 3435), yannick.lhorty@u-pem.fr.

1. Introduction

This paper evaluates the impact of practicing literacy skills and of measured improvement of these skills on the academic performance of first-year university students. Micro-econometric methods for policy, program and treatment effects are widely used to evaluate educational programs (Heckman and Rubinstein, 2001; Rivkin *et al.*, 2005; Heckman and Vytlacil, 2007; Dearden *et al.* 2009). On one hand, several articles already evaluate the impact of literacy on academic achievements. Most of these studies focus on elementary school and deal with the evaluation of educational programs implemented by governments to increase literacy skills, such as "No child behind" in the UK (since 2001), or "Success for All" in the US (since 2003). For instance, Machin and McNally (2008) evaluate the impact on academic achievement of elementary school students who participated in the program "the literacy hour" which is designed to support reading instruction in the UK.

On the other hand, few existing analyses focus on the relation between literacy skills and academic performance at the university level. Despite the limited data available, Delgadova (2015) reports that 40 percent of first-year university students suffer from severe problems in basic literacy skills and that these problems can be linked to their academic failure. The author relies on a qualitative analysis and descriptive statistics, but does not provide any empirical evidence for a causal effect. However, this lack of empirical evidence may deserve more attention in light of studies, such as Calmant and Hallier (2008) for France, reporting that students who leave the higher education system without receiving any degree may be more likely to be unemployed, compared to those who do finish a degree or other certificate of higher education.

In this paper, we evaluate the impact of practicing and improving literacy skills on the academic performance of first-year university students. For this purpose, we implement a randomized encouragement experiment (Duflo, Glennerster and Kremer, 2007) in a group of first-year university students. Our experiment involved all students entering the first year of university in Economics and Management over 2011-2014 at two French universities, Paris-Est Marne-La-Vallée and Lille 1. Each week of the first term of the academic year, half of the students were encouraged to improve their literacy skills through the use of an innovative educational tool, called *Projet Voltaire*, whereas the other half was not encouraged at all, although they also had access to this tool. Moreover, the Projet Voltaire allowed us to evaluate the literacy level of students at the beginning and at the end of the first term of the academic year, i.e. before and after the literacy training period. We consider a sample of 849 university students, for whom baseline administrative information is available, as well as scores on the two literacy tests, all final exam scores and first year grade averages. We then evaluate the impact both of the encouragement to practice literacy skills and of improving such skills on academic performance. Using local average treatment estimators (Imbens and Angrist, 1994; Angrist and Imbens, 1995; Angrist, Graddy and Imbens, 2000), we show that increasing literacy test scores has a positive impact on first-year academic results, whether we consider language-based disciplines or more scientific disciplines. In particular, depending on the kind of discipline we consider, learning literacy could increase academic test scores by a half-point to one-and-a-half points. Consequently, it increases the probability of first-year university students to complete the first or the second term, and even the full academic year. Finally, encouragement to practice literacy skills had an increased benefit on students who initially tested at lower levels for literacy skills. These findings hold for both universities, Paris-Est Marne-La-Vallée and Lille 1, although student populations at the two universities are quite different.

To our knowledge, this study is the first to use a randomized experiment to analyze the role played by literacy as a driver of academic success in university students. Our results suggest that public policy goals should include improving literacy skills among university students. In particular, our findings provide additional empirical support for educational programs that have already been implemented in some countries. The remainder of the paper is organized as follows. Section 2 describes the randomized experiment design we implemented to encourage students to practice literacy skills. Section 3 presents the empirical strategy used to identify the effects of improved literacy on academic performance of first-year university students. Section 4 discusses results. Section 5 provides a check on the robustness of the results and Section 6 concludes with recommendations.

2. A randomized encouragement experiment on literacy skills practice

The randomized experiment was first carried out at University Paris-Est Marne-La-Vallée (Paris region, France). It was implemented during tutorial classes whose goal is to teach methodological skills to all first-year university students in Economics and Management. Similar tutorials take place in all French universities. The experiment began at the beginning of the first semester of the academic year, when the instructor would tell the students about an online tool – called *Projet Voltaire* – that they could use to practice literacy skills (orthography, grammar, conjugation and syntax). This tool provides seven ordered levels of exercises for improving literacy skills and includes an application that keeps track of recurring mistakes for a given user profile, thus allowing students who practice to increase their literacy skill level. Once this initial information was given to students, they received differentiated information, according to the tutorial group they belong to.

In half of the tutorial groups, teachers actively encouraged their students to use the *Projet Voltaire* tool more intensively, by following a precise and identical protocol for each of these groups. At each tutorial session, they reminded students about the importance of literacy skills, or spent a few minutes reviewing a few relevant tips and rules. They also gave detailed explanations on how to use the *Projet Voltaire* web platform and reminded students that a test score would be given for this work and would be taken into account to compute final grades for the methodology course unit. We refer to these tutorial groups as the "encouraged" groups. In the other half of tutorial groups, nothing specific was done to encourage students to practice literacy skills via the *Projet Voltaire* web platform following the initial announcement given to all students; we refer to these tutorial groups as "non-encouraged". The students were divided and tutorial groups formed based simply on alphabetical order, and

teachers were randomly assigned to tutorial groups. This assures that the assignment of students to the encouragement condition was random.

This trial was first implemented with first-year students in Economics and Management at University of Paris-Est Marne-La-Vallée for three consecutive academic years between 2011-2012 and 2013-2014. Our final sample includes 526 students entering the literacy skills training experiment for the first year (2011), for whom information from a baseline administrative survey -necessary to the evaluation - is available. The average age of the first-year university student population is 18.5. More than half are male (56.8%) and almost all are of French nationality (94.5%). They are frequently scholarship students (37.6%). In addition, they often hold a baccalaureate with a concentration in Economics and Social Science (58.6%), with a Scientific profile (29.5%), or in Sciences and Technology in Management (8%). Most live in the region around Paris, in one of several administrative departments neighboring the University area, such as Seine-et-Marne (55.6%), Seine-Saint-Denis (24.7%) or Val-de-Marne (15.2%). Moreover, just over half of them passed the baccalaureate school-leaving exam in the Seine-et-Marne department (50.4%).

This random trial was extended in 2013-2014 to first-year university students in Economics and Management at Lille University of Science and Technology (Université Lille 1), with some minor differences in implementation due to the local organization of teaching. In this portion of the study, we retained a final dataset of 323 first-year university students. This complementary random experiment allows us to verify the external validity of the results we obtained through the main experiment at University Paris-Est Marne-La-Vallée.

Our evaluation of the impact of literacy skills training on student academic achievement relies on several types of information. First, we needed access to complete information about the first and second term grades in all disciplines for the entire first-year student group in Economics and Management. The academic performance data were provided by official grade reports for the two semesters by both universities. For every student and course taken, these reports contain final grades as well as grades for the final exam and the in-class assessment component that together make up the overall course grades. In addition to the course grades, they also include the student's overall average in all courses for the semester. Next, we merged this information with data provided by the web platform tool Projet Voltaire, specifically extracting the scores from the two literacy tests that all students took before and after the literacy practice period. Table 1 shows that the score of the initial literacy test⁴ for first-year students in Economics and Management at UPEM is on average 6.1 points (on a scale of 20). This average level is the same for the two categories of tutorial groups, which also showed no significant differences regarding variables provided by the following baseline administrative information: age, gender, nationality, type of baccalaureate, French administrative department where baccalaureate was awarded, current department of residence,

_

⁴ The time allowed for each of the two literacy tests is at most 45 minutes. These tests evaluate literacy skills that are supposed to be acquired by students who hold a baccalaureate.

and scholarship status. This table confirms that students were randomly assigned to the groups, independently of the type of encouragement.

Table 1. Testing for differences in sample characteristics between tutorial groups that were encouraged to practice literacy skills and those who were not; all are first-year university students in Economics and Management at University Paris-Est Marne-La-Vallée.

Characteristic	Encouraged (1)	Not encouraged (2)	Difference (1)-(2) (significance)
Score on first literacy test ^a	6.5	6.4	0.1
Age ^b	18.6	18.4	0.2
Gender (% male) ^c	56.5	57.3	-0.7
French nationality	94.3	94.7	-0.3
Scholarship student	37.8	37.4	0.4
<i>Type of baccalaureate</i> ^c :			
Bac ES (Economics and Social Science track)	56.8	60.8	-2.0
Bac S (Science track)	30.0	28.6	1.4
Bac STG (Technology in Management track)	9.4	6.2	3.2
Other (Literary track; foreign student)	3.7	4.4	-0.7
French department for baccalaureate c:			
Seine et Marne	50.2	50.7	-0.5
Seine-Saint-Denis	29.8	27.3	2.4
Val de Marne	10.4	13.2	-2.8
Other (including in the provinces or abroad)	9.3	0.4	-0.9
French department of residence ^c :			
Seine et Marne	54.5	56.8	-2.3
Seine-Saint-Denis	25.1	24.2	0.9
Val de Marne	15.4	15.0	0.4
Other (including in the provinces or abroad)	5.0	4.0	1.0

Source: randomized experiment implemented at University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year students in Economics and Management entering the university in 2011-2013, for whom information from the baseline administrative survey is available, in addition to scores for the two literacy tests.

Notes: a score; b years; c percentage. *** (respectively ** or *) stands for significance of the difference at a 1% level (5% or 10% respectively).

Third, we consider data provided by the pedagogical tool *Projet Voltaire*. In fact, *Projet Voltaire* provides us the literacy practice time data over the experimental period (between the first and the second literacy tests), as well as the overall connection time to the web platform. These connection times include the time needed to complete the two literacy tests, as well as the time used by students to do training exercises or to read and learn specific rules in orthography, grammar, conjugation or syntax.⁵ Consequently, we have two means of measuring time spent practicing literacy skills: time spent on individual training exercises and overall training time, which is the difference between the overall connection time and time spent to complete the two tests.

Such information allows us to test the efficiency of the encouragement device. Whatever the given econometric specification or the considered indicator used to measure literacy skills

⁵ The overall connection time also includes the time during which a student is connected without doing any exercises or studying literacy rules. However, this time is rather short because any user connected to the web-platform is automatically disconnected after more than a few minutes of inactivity.

practice time, Table 2 shows a positive correlation between encouragement and practice time. In other words, students who had benefited from encouragement spent more time practicing - 59 to 79 minutes more over the experimental period - than other students.

Table 2. Measured effect of receiving active encouragement on the literacy practice time for first-year university students in Economics and Management at University Paris-Est Marne-La-Vallée, considering different econometric

specifications.

Literacy practice time indicator (specification)/	Indicator 1	Indicator 1	Indicator 1	Indicator 2	Indicator 2	Indicator 2
Explanatory variables	(1)	(2)	(3)	(1)	(2)	(3)
Encouragement	58.397***	58.562***	58.989***	78.801***	78.672***	79.146***
	(11.258)	(11.315)	(11.088)	(15.300)	(15.354)	(14.934)
Score on first literacy test	, ,	-1.984	-6.869***	, ,	1.555	-5.045
•		(2.347)	(2.482)		(3.245)	(3.423)
Age			-0.439			-0.502
			(1.832)			(2.411)
Gender (% male)			-68.015***			-94.006***
			(12.287)			(16.497)
Scholarship student			18.374			26.752
•			(12.646)			(17.014)
Scientific Baccalaureate			11.720			18.671
			(11.967)			(16.299)
			-33.831			-43.384
			(22.258)			(26.476)
Other baccalaureate			-55.644**			-67.836**
			(22.798)			(28.668)
Intercept	71.983***	84.816***	157.745***	107.493***	97.437***	193.506***
•	(7.095)	(17.262)	(40.886)	(9.982)	(23.341)	(54.727)
Observations	526	526	526	526	526	526
R2	0.045	0.046	0.121	0.045	0.045	0.123
F	26.91	13.50	11.34	26.53	14.18	12.35

Source: randomized experiment implemented at University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year students in Economics and Management entering the university in 2011-2013, for whom information from the baseline administrative survey is available, in addition to scores for the two literacy tests.

Notes: effect of encouragement on time to literacy practice (OLS estimator). Robust standard errors within parentheses. Time to practice literacy: indicator 1 = indicator provided by platform Projet Voltaire; indicator 2 = overall time spent using platform Projet Voltaire - duration of the 1st literacy evaluation - duration of the 2nd literacy evaluation. Explanatory variables: score on the first literacy test, student age; student gender (reference=female); scholarship student; baccalaureate (reference= baccalaureate ES, Economics and Social Science track); baccalaureate with merit (reference= baccalaureate without merit). Robust standard errors. *** (respectively ** or *) stands for significance at a 1% level (respectively a 5% or 10%).

Reading: at a 1 percent level, and considering the first time indicator for literacy training, encouraging students to practice literacy skills implies an increase in literacy practice time of about one hour for first-year students in Economics and Management at University Paris-Est Marne-La-Vallée.

Finally, at the end of the first semester and before final exams took place, the period for practicing literacy skills was closed and a second literacy test was administered to all students. This second literacy test allows us to build an indicator to measure the increase in literacy skills, namely the difference between the scores on the two literacy tests. Tables A1 and A2 in the appendix show a positive correlation between the increase in literacy test scores and the amount of time students spent practicing literacy skills.

3. Identification

We use our randomized experiment to evaluate the impact of literacy skills on the academic performance of first-year university students in Economics and Management.

Using the framework of the Rubin Causal Model (Rubin, 1974), hereafter RCM, we identify the causal effect using an instrumental variable estimation strategy, following Angrist, Imbens and Rubin (1996). Y_i is the outcome variable. It represents the grade for the final exam, the inclass component or the overall course grade for any course taken by student i in the first year of the Economics and Management program; it may also refer to whether student i completes the academic period in question (i.e., the first or second semester or the full academic year). T_i is the treatment, and refers to varying the literacy skill level as measured by the two literacy test scores. We first consider T_i as taking only two values: either there is an increase in the literacy level (T_i =I, literacy score is greater for the second than for the first test) or not (T_i =0). We then evaluate the impact of increasing the literacy level on first-year students' final exam scores or final grades (for instance) in Economics and Management (average treatment effect):

$$ATE=E(Y_{1i} - Y_{0i})$$

Where Y_{Ii} (or respectively Y_{0i}) is the final exam score in the discipline in question if there is an increase in literacy test scores (or in absence of any increase, respectively). However, the treatment – increasing literacy skills or not – is endogenous: there are a lot of observed and unobserved variables that affect both the outcome and treatment variables. Hence, T_i is not randomly assigned. One way to identify ATE is to consider instrumental variables estimation. We thus need an instrument Z_i , *i.e.* a variable that is correlated with T_i , but not with the error term of the equation that describes outcome Y_i . Finding such a variable is difficult.

A randomized experiment makes it possible to solve this problem (Duflo *et al.*, 2007). Indeed, the encouragement design that was adopted in our experiment provides a valid instrument for increasing the literacy test score (or not) between the beginning and the end of the first term. In other words, whether a student is encouraged to practice literacy skills (Z_i =1) or not (Z_i =0) is assigned randomly.

Nevertheless, within the framework of the RCM and as shown in Angrist *et al.* (1996), the instrumental variables estimator identifies the effect of increasing literacy skills only for compliers, *i.e.* the university students whose probability of improving their literacy skill level increases if they are encouraged, in comparison to a situation where they are not encouraged. This refers to the *local average treatment effect* (Imbens and Angrist, 1994):

$$LATE = E(Y_{1i} - Y_{0i} | c_i) = \frac{E(Y_i | Z_i = 1) - E(Y_i | Z_i = 0)}{E(T_i | Z_i = 1) - E(T_i | Z_i = 0)}$$

Where c_i corresponds to students for whom the treatment effect is computed, i.e. those who comply with the encouragement and increase their time practicing literacy skills if they are in the encouraged group, and $Pr(T_1 - T_0) = E(T_i | Z_i = 1) - E(T_i | Z_i = 0)$ is the percentage of compliers in the whole population (T_1) is the value of treatment if Z=1 and T_0 is that corresponding to Z=0). Finally, $E(Y_i | Z_i = 1) - E(Y_i | Z_i = 0)$ is the effect of encouragement to practice literacy skills on the academic performance of first-year university students in Economics and Management (intention to treat). LATE responds to the following questions: by how many points does a student's final grade for a given course increase if the student's literacy skills improve? What is the impact on the probability for students to complete the first, the second term or the whole academic year if literacy levels are raised?

Up until now, we have considered a binary treatment (i.e., improving literacy skills or not). In practice, however, what we observe is the variation in the two literacy test scores for each student, which means the treatment has a variable intensity. According to Angrist and Imbens (1995) who considered a discrete ordered treatment, the local average treatment effect is still the following:

$$LATE = \frac{E(Y_i|Z_i = 1) - E(Y_i|Z_i = 0)}{E(T_i|Z_i = 1) - E(T_i|Z_i = 0)} = \sum_{j=1}^{J} \omega_j E(Y_j - Y_{j-1}|T_1 \ge j > T_0)$$
 with $\omega_j = \frac{\Pr(T_1 \ge j > T_0)}{\sum_{j=1}^{J} \Pr(T_1 \ge j > T_0)}$

In this case, for the final grade in a given course, the Wald estimator is an average of effects evaluated for different improvement (variation) intensities in the literacy test scores, weighted by the percentage of the relevant population for the given treatment intensity. Angrist, Graddy and Imbens (2000) generalize this result in the case of a continuous treatment. LATE then corresponds to the effect of a variation in literacy skills on academic performance for the university students in our study: by how many points does a final grade for a given course increase if the variation in the literacy test scores increases by 1 point? What is the impact on the probability for students to complete the first or the second semester if literacy levels are raised, and by how many percentage points does this probability increase or decrease?

4. Effects of improving literacy on academic performance in university students

In this section, we begin by presenting results from the randomized experiment that was implemented at University Paris-Est Marne-La-Vallée. Next, we estimate the effects of practicing literacy skills for different subpopulations.

4.1. First results

The main results of our randomized trial are presented in Table 3a and 3b. Effects of both encouragement and measured improvement in literacy skills are reported.⁶

First of all, encouraging students to practice literacy skills raises their scores in several disciplines, ranging from language-based disciplines (e.g., English as a foreign language), to somewhat formalized subjects (introduction to economics or to management), and even very formal subjects (introduction to microeconomics; mathematics or statistics). Final exam scores and final course grades are particularly impacted by literacy encouragement (Table 3a).

Second, the same kinds of results are observed when considering the effect of increased scores on the literacy test. More precisely, improving a student's literacy test score by one additional point implies an increase of 0.5 to 1.0 point (on a scale of 20) in the final exam score, the in-class assessment component or the final grade for a given course. This increase is larger for more scientific or formalized disciplines (0.8-1.0 point) than for other subjects (0.5-0.6 point). For instance, one additional point in the improvement measured between literacy test scores raises the final exam score in mathematics by 0.836 point, and the final grade in microeconomics by 0.769 point, whereas it raises the final exam grade in introduction to economics by 0.552 point and the in-class grade component in management by 0.494 (Table 3a). We obtain similar results if we add baseline administrative variables as explanatory variables, with results that are even more frequently significant.⁷

_

⁶ Only significant results are displayed in Table 3. Tables A3 to A6 in the appendix include detailed complete results.

⁷ This result was expected because adding relevant explanatory variables reduces the residual sum of squares and thus the standard errors of the estimated coefficients. In particular, at a 10 percent level, one additional point in the improvement measured between literacy test scores would increase the average grade for module 1 of semester 2 by 0.499, and increase Semester 2 grade averages by 0.287 point.

Table 3a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée.

Discipline	Introduction to economics (FE)	Introduction to management (CA)	Mathematics (FE)	Introduction to Microeconomics (GPA)	Introduction to Microeconomic(CA)	Introduction to Microeconomics(FE)	Financial economics (GPA)	Financial economics (CA)	Statistics and computer science (GPA)	Statistics and computer science (FE)	English (Semester 2) (GPA)
Model				Ef	fect of encouragement to li	teracy learning (Intention	To treat, ITT)				
Without baseline variables	0.708** (0.335)	0.642*** (0.215)	1.067** (0.443)	1.082***(<i>a</i>) (0.406)	0.838** (0.392)	0.665 (0.413)	0.587 (0.385)	0.607 (0.379)	0.875** (0.369)	0.448 (0.396)	0.384* (0.230)
With baseline variables	0.683** (0.313)	0.627*** (0.206)	1.036*** (0.387)	1.048*** (0.367)	0.954*** (0.366)	0.828** (0.396)	0.619* (0.353)	0.728** (0.360)	0.807** (0.324)	0.641* (0.365)	0.477** (0.206)
Modèl				Effect	of varying literacy test scor	es (Local Average Treatm	ent Effect, LAT	E)			
Without baseline variables	0.581** (0.227)	0.511*** (0.173)	0.879***(<i>b</i>) (0.343)	1.032*** (0.376)	0.845** (0.402)	0.690* (0.420)	0.558* (0.337)	0.578* (0.352)	0.840** (0.335)	0.391 (0.519)	0.600* (0.353)
With baseline variables	0.552** (0.253)	0.494*** (0.167)	0.836*** (0.298)	0.997*** (0.356)	0.867** (0.353)	0.769** (0.368)	0.573* (0.308)	0.640** (0.316)	0.769*** (0.297)	0.607* (0.330)	0.479** (0.231)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014) and Tables A3 to A6 (appendix).

Field: 526 first-year students in Economics and Management entering the university in 2011-2013, for whom information from the baseline administrative survey is available, in addition to scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.082 points the score to the grade point average in introduction to microeconomics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.879 point in the score of the final exam in Mathematics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 3b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Marginal effects*.

		E	0 33	
Discipline	Academic	First	Second	
	Year	Semester	Semester	
Model	Effect of encour	agement to literacy learning (Inter	ntion To treat, ITT)	
Without	0.070	0.056	0.083*	
baseline variables	(0.137)	(0.214)	(0.071)	
With	0.088**(a)	0.062	0.086*	
baseline variables	(0.040)	(0.116)	(0.051)	
Model	Effect of varying lite	racy test scores (Local Average Ti	reatment Effect, LATE)	
Without	0.068**	0.045	0.071***	
baseline variables	(0.024)	(0.151)	(0.002)	
With	0.073***(b)	0.048**	0.068***	
baseline variables	(<0.001)	(0.041)	(0.003)	

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 8.8 percentage points the probability to achieve first-year university for students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 7.3 percentage points the probability to achieve first-year university for students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France)

Finally, the above findings suggest greater chances of students completing the first year of university. As shown in Tables A3-A6 in the appendix, first-year grade averages do not change through the implementation of the encouragement device to practice literacy skills. However, Table 8b shows that encouraging students to practice literacy skills increases the probability of their completing the second term of the first year by 8.5 percentage points, and the full year by 8.8 percentage points, although there is no observable effect on the probability to complete the first term. Further, each additional point in the improvement measured between literacy test scores raises the probability for the students to complete their first year at university by 6.8 to 7.3 percentage points.

4.2 Heterogeneous treatment effects

In Section 4.1, we show that raising literacy skills helps in increasing achievement test scores for first-year university students in Economics and Management stream. These impacts can differ if we consider different subpopulations. In fact, Table 4 shows that initial literacy test score is not the same for some subpopulations of students. In particular, initial literacy test score is on average smaller among male students than among female students (6.0/20 vs. 7.2/20). As well, initial literacy test score is on average smaller among students who hold a baccalaureate without any merit, honors or distinction that among other students (6.0/20 vs. about 7.4/20). Finally, there seems to be no differences in average initial literacy test scores between students whose mother tongue is the French language and other students. Hence, it may be interesting to see whether or not the encouragement device is more efficient for some subpopulations than for others.

Table 4. Score to the first written literacy test for first-year university students in Economics and Management at university Paris-Est Marne-La-Vallée. *Considering different kinds of populations*.

Sub-sample Sub-sample	Number of students	Average	Standard deviation
All students	526	6.5	2.4
Gender:			
Men	299	6.0	2.3
Women	227	7.1	2.5
Origin:			
French	315	6.6	2.4
Foreign	168	6.4	2.5
Baccalauréat:			
With merit, honours or distinction (pass 60%-70%, pass 70%-80%, pass 80% upwards	246	7.4	2.4
Pass 50%-60%	182	6.0	2.1
Repeat session	74	5.6	2.3
Repeat a year	24	6.1	2.6
Pass 50%-60%, repeat session, or repeat a year)	280	5.9	2.2

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Note: score.

First, Tables 5a and 5b show that positive impacts of encouragement and of increasing literacy skills on academic performance are mainly due to improvements among male students. The effects on female students are much smaller. Thus, the encouragement device seems to have preferentially benefited students whose literacy skills were initially weaker, *i.e.* male students.

Second, if we distinguish students according to their native language (French in contrast to any other language), we see that both encouragement and increasing literacy skills have different impacts on these subpopulations. More precisely, positive effects were larger for students whose mother tongue is French; moreover, they do not involve the same disciplines (Tables 6a and 6b).⁹

Third, Tables 7a, 7b, and 7c display results for three different subpopulations of students, depending on whether their baccalaureate was awarded with "merit, honors or distinction", or awarded without any distinction or not awarded. Although students who hold a baccalaureate without honors demonstrate lower initial literacy skills than other students, the increase in literacy test scores has a positive impact on academic performance for the three subpopulations. However, positive effects do not involve the same disciplines and they are somewhat larger for students who earned their baccalaureate without honors than for other students.¹⁰

Fourth, as Tables 5, 6 and 7 demonstrate, certain effects are actually negative, and may appear counter-intuitive given the overall impacts discussed above. For instance, for female students specifically, benefiting from encouragement negatively impacts grades in Management for example (considering the final exam and the final course grade). This pattern may be interpreted as a lock-in effect: when students spend time practicing literacy skills, they do not

⁸ Regardless of the experimental conditions received by students, female students spent more time practicing literacy skills than male students (161 to 230 minutes *vs.* 105 to 150 minutes, depending on the indicator used to measure practice time). Literacy test scores also increased more for female students than for males (+4.0 *vs.* +2.5 points). However, as a consequence of encouragement, literacy practice time increases more among males than among female students (+65 to +92 minutes *vs.* +48 to 60 minutes, depending on the indicator used for practice time). Hence, literacy test score increased more among male than among female students (+1.4 *vs.* +1 point) for students assigned to encouragement. Detailed results for male and female students and all disciplines are available on request.

While students whose native language is not French spent more time practicing literacy skills than native French-speaking students when neither benefited from any encouragement, practice time increases more among students whose first language is not French language than among native French-speaking students (+97 to +113 minutes vs. +38 to +60 minutes, depending on the indicator used). In spite of this, the increase in literacy test scores was only slightly larger among students whose first language is not French language (+3.6 vs. +3.1 points). Detailed results for these two types of populations and all disciplines are available on request.

Regardless of the experimental conditions they were assigned to, students who held a baccalaureate with merit,

Regardless of the experimental conditions they were assigned to, students who held a baccalaureate with merit, honors or distinction spent more time practicing literacy skills. Thus literacy test score increases more for these students than for others (+3.8 vs. +2.6 points). Nevertheless, time practicing literacy skills increases more among students who earned a baccalaureate without honors than among other students (+68 to 95 minutes vs. +46 to 57 minutes, depending on the indicator used). Hence, when both categories of students are assigned to encouragement, literacy test scores increased more among students who hold a baccalaureate with a lower score than among those holding a baccalaureate with merit, honors or distinction (+1,5 vs. +0,7 point). Detailed results for all disciplines are available on request.

spend the time on their homework in other subjects (like Economics or Management). Thus, for some disciplines, the encouragement device could have led to some negative effects. However, these results are often barely significant (here, at a 10 percent level), particularly when considering ** the effects of improving literacy test scores**. Hence, lock-in effects may have appeared, but their relevance seems to be very limited.

Table 5a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: male students*.

Discipline	Introduction	Introduction	Introduction	Mathematics	Mathematics	Introduction to	Introduction to	Introduction to	Economics and	Economics and	Economics and	General firm policy	Statistics a
	to Economics (FE)	to management	to management	(GPA)	(FE)	Microeconomics(GPA)	microeconomics(CA)	macroeconomics	Finance (GPA)	finance	finance	(F)	computer scie
		(GPA)	(CA)					(GPA)		(CA)	(FE)		(F)
Model						Effect of end	couragement to literacy l	earning (Intention	To treat, ITT)				
Without	0.777*	0.803***	1.017***	0.860	1.420***(a)	1.440***	0.870*	0.946*	1.258**	0.950*	0.584	0.859**	1.243**
baseline	(0.335)	(0.304)	(0.287)	(0.542)	(0.406)	(0.525)	(0.510)	(0.500)	(0.519)	(0.488)	(0.487)	(0.429)	(0.529)
variables													
With baseline	0.727***	0.861***	1.009***	0.920**	1.471***	1.428***	1.014**	0.871**	1.270***	1.000**	0.772*	0.794**	1.100***
variables	(0.283)	(0.297)	(0.291)	(0.458)	(0.525)	(0.484)	(0.495)	(0.446)	(0.480)	(0.490)	(0.466)	(0.397)	(0.442)
Model						Effect of varying	literacy test scores (Loc	al Average Treatm	ent Effect, LATE)				
Without	0.585*	0.569***(b)	0.723***	0.618*	1.051**	1.249***	0.832*	0.801**	1.046**	0.828*	0.570	0.731**	1.077**
baseline	(0.342)	(0.211)	(0.228)	(0.365)	(0.425)	(0.451)	(0.484)	(0.393)	(0.423)	(0.427)	(472)	(0.353)	(0.436)
variables													
With baseline	0.617**	0.591***	0.702***	0.659**	1.072***	1.243***	0.929**	0.750**	1.051***	0.848**	0,716*	0.687**	0.970***
variables	(0.314)	(0.202)	(0.220)	(0.310)	(0.374)	(0.439)	(0.469)	(0.363)	(0.404)	(0.418)	(0,440)	(0.339)	(0.384)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 299 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively at 5% or 10%) level.

Reading: (a) at a 1 percent level, encouraging to literacy practice using platform Projet Voltaire increases by 1.420 point the score to the final exam in mathematics for first-year university male students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris reg

percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.569 point in the score to the final exam in introduction to management for first-year university students in Economics and Management stream at university Paris-Est Marne-L

Table 5b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: female students*.

Discipline	Introduction to Management (GPA)	Introduction to Management (FE)	Methodology (CA)	Introduction to Microeconomics (FE)	Statistics and computer science (CA)
Model		(IL)	Effect of encouragement to literacy l		(CH)
Without baseline variables	-0.666**(a)	-1.437***	0.410	0.827	0.665*
	(0.329)	(0.444)	(0.283)	(598)	(0.397)
With baseline variables	-0.698**	-1.511***	0.450*	1.058*	0.654*
	(0.297)	(0.415)	(0.297)	(0.615)	(0.386)
Model		Ef	fect of varying literacy test scores (Loc	al Average Treatment Effect, LATE)	
Without baseline variables	-0.661	-1.426*(b)	0.394	0.835	0.606
	(0.472)	(0.796)	(0.283)	(0.635)	(0.395)
With baseline variables	-0.669	-1.449*	0.419	0.967	0.553
	(0.433)	(0.752)	(0.260)	(0.603)	(0.345)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 227 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** and *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice using platform *Projet Voltaire* decreases by 0.666 point the score to the final exam in introduction to management first-year university female students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a decrease of 1.426 point in the score to the final exam of introduction to management for first-year university female students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 6a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: students whose country of origin is France.*

Discipline	Introduction to Management	Mathematics (FE)	Dissertation	Introduction to Microeconomics(GPA)	Introduction to Microeconomics(CC)	Introduction to Microeconomics(FE)	Financial economics	Principles Of law	English language (Semester 2)
	(CA)						(CA)		
Modèl				Effect of encourageme	nt to literacy learning (Int	ention To treat, ITT)			
Without	0.730**	1.084*(a)	-0.463*	1.001*	0.831*	0.619	0.630	0.279	0.252
baseline	(0.267)	(0.570)	(0.255)	(0.510)	(0.490)	(0.504)	(0.472)	(0.294)	(0.278)
variables									
With baseline	0.738***	1.186**	-0.370*	1.138***	1.155***	0.954**	0.894**	0.548*	0.419*
variables	(0.263)	(0.498)	(0.207)	(0.453)	(0.456)	(0.476)	(0.448)	(0.301)	(0.254)
Modèl				Effect of varying literacy te	st scores (Local Average	Treatment Effect, LATE)			
Without	0.790***(b)	1.245*	-0.530	1.269*	1.105	0.843	0.752	0.409	0.362
baseline	(0.354)	(0.652)	(0.404)	(0.671)	(0.722)	(0.704)	(0.568)	(0.434)	(0.416)
variables									
With baseline	0.727***	1.197**	-0.380	1.249**	1.185**	0.968*	0.881*	0.590*	0.461
variables	(0.305)	(0.522)	(0.270)	(0.557)	(0.561)	(0.536)	(0.472)	(0.347)	(0.312)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 315 first-year university students in Economics and Management entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 10 percent level, encouraging to literacy practice increases by 1.084 point the score to the final exam in Mathematics microeconomics for (originating from France) first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.790 point in the score to continuous assessment in introduction to management for (originating from France) first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France)

Table 6b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: students whose country of origin is NOT France.*

Discipline	Introduction to Economics	Introduction to Microeconomics	Statistics and Computer science(GPA)	Statistics and Computer science(FE)	Principles of law	English language (Semester 2)
	(FE)	(GPA)	computer serence(GTTT)	computer science(12)	01 14.11	,
Model			Effect of encouragement to	literacy learning (Intentio	n To treat, ITT)	
Without baseline	1.643**(a)	1.068	1.413**	1.355**	-0.811*	0.704*
variables	(0.606)	(0.725)	(0.638)	(0.684)	(0.349)	(0.422)
With baseline variables	1.503***	0.692	1.064*	1.313**	-0.792*	0.801**
	(0.568)	(0.661)	(0.553)	(0.626)	(0.351)	(0.345)
Model		Effe	ect of varying literacy test so	cores (Local Average Treat	ment Effect, LATE)	
Without baseline	0.966***	0.653*	0.868**	0.932**(b)	-0.576*	0.528
variables	(0.367)	(0.392)	(0.349)	(0.430)	(0.309)	(0.334)
With baseline variables	0.959***	0.462	0.705**	0.908**	-0.580*	0.624*
	(0.383)	(0.397)	(0.334)	(0.402)	(0.307)	(0.327)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 168 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (*intention to treat*; OLS estimator) or of varying literacy test score (*local average treatment effect*; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. **** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice using platform *Projet Voltaire* increases by 1.643 point in the score of the final exam in introduction to economics for first-year university students (not originating from France) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 5

percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.932 point in the score of the final exam in statistics and computer science for first-year university students (not originating from France) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 7a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: students who hold a baccalaureate with merit, honors or distinction.*

Discipline	Introduction to Management (CA)	Mathematics (FE)	Methodology (CA)	Introduction to Microeconomics (GPA)	Introduction to Microeconomics (CA)	Introduction to Microeconomics (CT)	Statistics and computer science(GPA)	Statistics and computer science (CA)	Statistics and computer science (FE)	English language (Semester 2)
Model				Effect of en	couragement to literac	y learning (Intention To	treat, ITT)			
Without	0.557*	1.070*	0.471*	1.512**	1.459**	1.533**(a)	1.246**	0.828*	1.023*	0.652**
baseline	(0.297)	(0.643)	(0.277)	(0.607)	(0.568)	(0.624)	(0.525)	(0.426)	(0.588)	(0.299)
variables										
With baseline	0.569**	0.923*	0.542**	1.515***	1.376***	1.581***	1.106**	0.721*	1.023*	0.649**
variables	(0.282)	(0.571)	(0.262)	(0.569)	(0.540)	(0.615)	(0.485)	(0.397)	(0.551)	(0.271)
Model				Effect of varying	g literacy test scores (L	ocal Average Treatmen	t Effect, LATE)			
With baseline	0.594*(b)	1.153	0.493*	2.305	2.054	2.461	1.919	1.174	1.676	1.072
variables	(0.339)	(0.718)	(0.296)	(1.406)	(1.266)	(1.646)	(1.193)	(0.837)	(1.261)	(1.876)
Avec	0.589*	0.955*	0.533**	2.245*	1.782*	2.159*	1.650*	0.901	1.373	0.956
variables	(0.321)	(0.586)	(0.274)	(1.286)	(1.011)	(1.214)	(0.953)	(0.596)	(0.851)	(0.684)
baseline										

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 246 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.533 point the score to the final exam in introduction to microeconomics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.594 point in the score of the continuous assessment in introduction to management for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 7b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: students who hold a baccalaureate without merit, honors or distinction.*

Note	Introduction to Economics (FE)	Introduction to Management (CA)	Mathematics (GPA)	Mathematics (FE)	Introduction to Microeconomics(GPA)	Financial economics (GPA)	Financial economics (CA)
Model			Effec	et of encouragement to l	literacy learning (Intention To	treat, ITT)	
Without	1.159***(a)	0.723**	1.098*	1.098*	0.777	0.818*	0.712
baseline variables	(0.423)	(0.287)	(0.571)	(0.571)	(0.493)	(0.470)	(0.460)
With baseline	1.117***	0.700***	0.727*	1.167**	0.823*	0.848*	0.758*
variables	(0.397)	(0.288)	(0.436)	(0.502)	(0.443)	(0.442)	(0.463)
Model			Effect of	varying literacy test sco	ores (Local Average Treatmen	t Effect, LATE)	
Without	0.788***	0.468**	0.457	0.741**(b)	0.559*	0.596*	0.528
baseline variables	(0.319)	(0.193)	(0.320)	(0.358)	(0.336)	(0.331)	(343)
With baseline	0.736***	0.449**	0.469*	0.760**	0.585*	0.600**	0.545*
variables	(0.285)	(0.188)	(0.267)	(0.311)	(0.310)	(0.307)	(0.333)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 280 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is an am; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 1 percent level, encouraging to literacy practice increases by 1.159 point the score to the final exam in introduction in economics for first-year university students in Economics and Management stream at university Paris-Est

Marne-La-Vallée (Paris region, France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.741 point in the score of the final exam in Mathematics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 7c. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Paris-Est Marne-La-Vallée. *Considered sample: students who hold a baccalaureate with pass* 50%-60%.

Discipline	Introduction to	Introduction to	Introduction to	Introduction to	Mathematics	Introduction to	Introduction to	Financial	Entrepreneurship
	Economics	Economics	Management	Management	(FE)	Microeconomics (GPA)	Macroeconomics	Economics (F)	(F)
	(GPA)	(FE)	(GPA)	(CA)			(F)		
Model				Effect of end	couragement to literacy	learning (Intention To treat, I'	lT)		
Without	0.600	1.033**	0.517	0.708**(a)	0.665	0.748	0.718	0.935*	0.758
baseline	(0.400)	(0.502)	(0.371)	(0.354)	(0.704)	(0.624)	(0.557)	(0.549)	(0.465)
variables									
With baseline	0.717	1.076**	0.578*	0.725**	1.008*	0.946*	0.955*	1.121**	0.825
variables	(0.374)	(0.476)	(0.354)	(0.353)	(0.615)	(0.549)	(0.498)	(0.524)	(0.450)
Model				Effect of varying	literacy test scores (L	ocal Average Treatment Effect,	LATE)		
Without	0.312	0.552**(b)	0.268	0.370**	0.355	0.422	0.404	0.525*	0.423*
baseline	(0.202)	(0.277)	(0.180)	(0.184)	(0.352)	(0.324)	(0.294)	(0.287)	(0.246)
variables									
With baseline	0.366**	0.553**	0.294*	0.375**	0.517*	0.530*	0.533**	0.617**	0.459*
variables	(0.186)	(0.257)	(0.171)	(0.180)	(0.292)	(0.290)	(0.270)	(0.276)	(0.243)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 182 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.033 point the score to the final exam in introduction to economics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.525 point in the score of continuous assessment in financial economics for first-year university students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Finally, the encouragement device and the increase in literacy test scores imply an increased probability of completing a given academic term for first-year university students who are initially characterized by low literacy skills (Tables 8a and 8b; Tables 9a and 9b; and Tables 10a to 10c).

Table 8a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Male students*.

Discipline	Academic	First	Second
•	Year	Semester	Semester
Model	Effect of encour	agement to literacy learning (Inten	ntion To treat, ITT)
Without baseline variables	0.149** (0.019)	0.104* (0.081)	0.089 (0.155)
With baseline variables	0.159***(<i>a</i>) (0.009)	0.119** (0.033)	0.090 (0.119)
Model	Effect of varying lite	racy test scores (Local Average Tr	reatment Effect, LATE)
Without baseline variables	0.100*** (<0.001)	0.068** (0.011)	0.075** (0.028)
With baseline variables	0.099***(<i>b</i>) (<0.001)	0.074** (<0.001)	0.074** (0.013)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 299 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 1 percent level, encouraging to literacy practice increases by 15.9 percentage points the probability to achieve first-year university for male students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 9.9 percentage points the probability to achieve first-year university for male students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France)

Table 8b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Female students*.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encou	ragement to literacy learning (Inter	ntion To treat, ITT)
Without	-0.002	0.008	0.078
baseline variables	(0.971)	(0.898)	(0.253)
With	0.012	0.004	0.090(a)
baseline variables	(0.836)	(0.945)	(0.173)
Model	Effect of varying lit	eracy test scores (Local Average T	reatment Effect, LATE)
Without	0.004	0.012	0.066*
baseline variables	(0.951)	(0.827)	(0.060)
With	0.016	0.008	0.067**(b)
baseline variables	(0.759)	(0.868)	(0.025)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 227 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference=Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice does not change the probability to achieve second term of firstyear university for female students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 6.7 percentage points the probability to achieve second term of first term university for female students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, Fran In particular, encouragement increases the probability for males students to complete the first year of university by 15.9 percentage points, and increases the probability for students whose country of origin is not France to complete the first term by 15.4 percentage points. ¹¹ Further, improving the literacy test score by 1 point increases the probability for male students to complete their first year of university by 10 percentage points. Also, for students from countries outside of France, one additional point in the improvement measured between literacy test scores increases the probability of completing the first semester by 8.2 percentage points. For students who **repeat the baccalaureate exam before successfully passing it**, the probability of completing their second semester of the first year of university is increased by 7.1 percentage points. ¹²

Table 9a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:*

students whose country of origin is France.

Discipline	Academic	First	Second
Discipline	Year	Semester	Semester
	i ear	Semester	Semester
Model	Effect of encour	agement to literacy learning (Inter	ntion To treat, ITT)
Without	0.046	0.027	0.075
baseline	(0.431)	(0.638)	(0.201)
variables			
With	0.085(a)	0.032	0.076
baseline	(0.112)	(0.538)	(0.159)
variables			
Model	Effect of varying lite	racy test scores (Local Average Tr	reatment Effect, LATE)
Without	0.059	0.035	0.078***
baseline	(0.287)	(0.564)	(0.006)
variables			
With	0.078***(b)	0.041	0.071**
baseline	(0.002)	(0.371)	(0.017)
variables			

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 315 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve first-year university for students (whose country of origin is France) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 7.8 percentage points in the probability to achieve first-year university for students (whose country of origin is France) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France)

To conclude, we have provided new empirical evidence that encouraging university students to practice literacy skills and increasing their literacy skill levels may help them in completing their first year at university.

-

¹¹ It also increases the probability of completing the first term by 10 percentage points for first-year university students holding a baccalaureate with distinction, honors or merit.

¹² Improving literacy performance by 1 point also increases the probability of completing the academic year (or at least one semester) for other types of students, namely those who hold the baccalaureate with honors or merit are more likely to complete the full year, while students of French nationality are more likely to complete the first term and female students are more likely to complete the second term.

Table 9b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample: students whose country of origin is NOT France.*

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encou	uragement to literacy learning (Intent	tion To treat, ITT)
Without	0.087	0.154**(a)	0.092
baseline variables	(0.290)	(0.046)	(0.244)
With	0.095	0.150**	0.087
baseline variables	(0.208)	(0.036)	(0.252)
Model	Effect of varying li	teracy test scores (Local Average Tre	eatment Effect, LATE)
Without	0.062	0.082***(b)	0.058
baseline variables	(0.181)	(0.001)	(0.142)
With	0.058	0.076***	0.052(b)
baseline variables	(0.153)	(0.001)	(0.220)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 168 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 level, encouraging to literacy practice increases by 15.4 the probability to achieve the first term in first-year university for students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 8.2 percentage points the probability to achieve the first-term of first-year university for students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, Fran

Table 10a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:* students who hold a baccalaureate with merit, honors or distinction.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encour	agement to literacy learning (Inter	ntion To treat, ITT)
Without	0.100*	0.063	0.137***
baseline variables	(0.068)	(0.288)	(0.032)
With	0.108**(a)	0.070	0.140***
baseline variables	(0.027)	(0.167)	(0.024)
Model	Effect of varying lite	racy test scores (Local Average Tr	reatment Effect, LATE)
Without	0.096***	0.059	0.091***
baseline variables	(<0.001)	(0.168)	(<0.001)
With	0.098***(b)	0.062**	0.095***
baseline variables	(<0.001)	(0.037)	(<0.001)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 246 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increase by 10.8 percentage points the probability to achieve first-year university for students (who hold a baccalaureate with merit, honors or distinction) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 9.8 percentage points in the probability to achieve first-year university for students (who hold a baccalaureate with merit, honors or distinction) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 10b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:* students who got a baccalaureate without merit, honors or distinction.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encour	ragement to literacy learning (Inten	tion To treat, ITT)
Without	0.060	0.050	0.048
baseline variables	(0.348)	(0.415)	(0.411)
With	0.083(a)	0.061	0.050
baseline variables	(0.150)	(0.287)	(0.374)
Model	Effect of varying lite	eracy test scores (Local Average Tr	reatment Effect, LATE)
Without	0.042	0.033(b)	0.035
baseline variables	(0.302)	(0.396)	(0.396)
With	0.052*(b)	0.039	0.032(b)
baseline variables	(0.076)	(0.235)	(0.398)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 280 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the first term in first-year university for students (who hold a baccalaureate without any merit, honors or distinction) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 5.2 percentage points the probability to achieve the first-term of first-year university for students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

Table 10c. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:* students who got a baccalaureate with pass 50%-60%.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encou	ragement to literacy learning (Inten	tion To treat, ITT)
Without	0.035	0.023*(a)	0.058
baseline variables	(0.660)	(0.077)	(0.430)
With	0.088	0.056	0.081
baseline variables	(0.232)	(0.431)	(0.249)
Model	Effect of varying lit	eracy test scores (Local Average Tr	eatment Effect, LATE)
Without	0.020	0.012(b)	0.033
baseline variables	(0.670)	(0.772)	(0.418)
With	0.047	0.029	0.042(b)
baseline variables	(0.173)	(0.424)	(0.230)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 182 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests. Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the first term in first-year university for students (who got a baccalaureate with pass 50%-60%) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At any level, an increase of 1 point in the difference between the second and the first literacy test scores let unchanged the probability to achieve the first-term of first-year university for students (who got a baccalaureate with pass 50%-60%) in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France).

5. Robustness: external validity

The results discussed above were obtained through a randomized controlled experiment and are therefore characterized by a high degree of internal validity. The increase in literacy skills often leads to a significant improvement in academic performances of first-year university students in Economics and Management. However, these results might somehow be specific to the University Paris-Est Marne-La-Vallée, where we carried out the experiment. For instance, they could be related to local characteristics, like the way courses are organized, or the socio-demographic characteristics of students. In order to establish the external validity of our results, we implemented the same kind of encouragement device over the academic year 2013-2014 at another French university, University Lille 1, again with first-year students in Economics and Management. This parallel experiment may help us to generalize the findings we obtained at the University Paris-Est Marne-la-Vallée.

5.1 A second randomized experiment at University Lille 1

As in the experiment at the University Paris-Est Marne-La-Vallée, the randomized trial was implemented at University Lille 1 over the first term of the academic year. During the meeting with students before the beginning of the academic year 2013-2014, students were informed that they would have access to the web platform Projet Voltaire to practice and improve their skills in orthography, grammar, conjugation and syntax. They were also told that their work would be evaluated through a final literacy test score that would be taken into account to compute their overall academic average for the first year in Economics and Management. Just as we did at the University Paris-Est Marne-La-Vallée, we evaluated initial literacy skills of students through an initial literacy test at the beginning of the academic year.

We also distinguished two categories of tutorial groups. In half of them, students were assigned to receive encouragement to practice literacy skills using Projet Voltaire, while the students of other groups were not encouraged. In the encouraged tutorial groups, e-mails were sent every week to the students, to remind them that they could use the Projet Voltaire platform and how to access this online tool. They were also reminded of the importance of using the platform to practice and increase their skills in orthography, grammar, conjugation and syntax. Finally, they were told that their work would be evaluated through a second literacy test that will be written at the end of the first term of the academic year, i.e. before final exams took place.

In the "non-encouraged" tutorial groups, nothing was said to students after the very first meeting at the beginning of the academic year, where they quickly heard about the Projet Voltaire. As with UPEM, tutorial groups were assigned randomly to the encouragement condition, and we confirmed this through t-tests to see whether or not there are any differences in baseline administrative variables (see Table A7 in appendix). Finally, as with UPEM, the difference between the two literacy test scores is used as a measure for the

increase in literacy skills of the students over the first term, i.e. between before and after the literacy practice period. 13

Our final sample includes 323 first-year students at University Lille 1, for whom we have the information needed to evaluate the effects of encouragement and of improving literacy skills (variables provided by baseline administrative information, the two literacy test scores and time practicing literacy skills, and final exam scores in first-year university). First-year students in Economics and Management at University Lille 1 are older (on average 19.5 years) than those who study at UPEM. In comparison with UPEM, they are also more frequently male students (65.8%) and less frequently of French nationality (80.8%). More of them are also scholarship students (51.1%). In addition, 50.3% of first-year university students at University Lille 1 hold a baccalaureate in Economics and Social sciences, 31.3% a baccalaureate in Sciences and 9.6% a baccalaureate in sciences and technology in Management. Not surprisingly, almost all students live in the Northern region of France, where the university is located (96.9 %). Despite these demographic differences, the average initial literacy test score is 6.1/20, which is identical to the average score for first-year university students in Economics and Management at UPEM.

5.2 Results

In Tables 11a and 11b, we present the main significant results of the effect of encouragement and of the increase in literacy test scores on academic performance for first-year students in Economics and Management at University Lille 1.¹⁴

Table 11a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management in university Lille 1.

u c u c c i i i i	perrormanee	Tot thist your c	in consider	onto in Economi	e united 1, 1 united	germent in ann versi	ey Ellie I.
Discipline	Introduction to Economics (GPA)	Introduction to Economics (CA)	Introduction to Management	Defining career objectives	Economic History	English language Semester 1 (CA)	Mathematics (CA)
Model			Effect of encourage	ement to literacy learni	ng (Intention To	treat, ITT)	
Without	0.751*	0.665	0.998*	1.606***	1.118*	1.705*	2.008***
baseline variables	(0.388)	(0.432)	(0.529)	(0.310)	(0.575)	(0.990)	(0.732)
With	0.747**	0.693*	0.938*	1.533***	1.225**	1.701*	2.486***
baseline variables	(0.359)	(0.399)	(0.521)	(0.305)	(0.532)	(0.946)	(0.698)
Model		Eff	ect of varying literac	y test scores (Local Av	verage Treatmen	t Effect, LATE)	
Without	0.460**	0.414	0.568*	0.974***	0.674**	0.993*	1.419**
baseline variables	(0.226)	(0.258)	(0.294)	(0.234)	(0.333)	(0.551)	(0.561)
With	0.486**	0.454*	0.563*	0.984***	0.780**	1.057*	1.789***
baseline variables	(0.220)	(0.248)	(0.307)	(0.247)	(0.327)	(0.560)	(0.577)

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014) and Tables A9 to A11 (appendix).

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.225 point the score to the final exam in economic history for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0,486 point in the GPA in economic history for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

¹³ We verify that students who were encouraged spent more time practicing literacy skills. Furthermore, there is a positive correlation between the increase in literacy test scores and time spent on literacy skills training. Corresponding tables are available on request.

¹⁴ Complete detailed results are found in the appendix (Tables A8 to A11).

Table 11b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Marginal effects*.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of en	couragement to literacy learning	(Intention To treat, ITT)
Without baseline	0.039	0.096*	0.009
variables	(0.490)	(0.085)	(0.865)
With baseline	0.048	0.100*(a)	0.016
variables	(0.351)	(0.052)	(0.748)
Model	Effect of varying	g literacy test scores (Local Aver	age Treatment Effect, LATE)
Without baseline	0.022	0.060**	-0.001
variables	(0.554)	(0.042)	(0.976)
With baseline	0.028	0.065**(b)	0.004
variables	(0.412)	(0.022)	(0.914

 $Source: randomized\ experiment\ implemented\ in\ University\ Lille\ 1\ (Northern\ from\ France;\ 2013-2014).$

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 10 percent level, encouraging to literacy practice increases by 10.0 percentage points the probability to achieve the first term of the first-year university for students in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 7.3 percentage points the probability to achieve first-year university for students in Economics and Management stream at university Lille 1 (Northern from France)

First, improving a student's literacy test score by one additional point between the beginning and the end of the first term implies an increase of 0.5 to 1.5 points in the grade for a given discipline (Table 8a).

Second, as with UPEM, the measured increase in literacy skills affects scores in a range of different disciplines, whether more literary or linguistic (e.g., English language, economic history, the "defining career objectives" workshop), or more technical (introduction to economics, introduction to management), and even in mathematics. Thus, one additional point in the improvement measured between literacy test scores induces an increase of 0.486 point in the final exam score in introduction to economics, or an increase of 0.454 point in the inclass assessment grade in introduction to management. The increase in grades is the largest for the mathematics in-class component (+1.789 points – Table 11a), followed by "defining career objectives" (+0.984 points).

Third, as in the case of University Paris-Est Marne-La-Vallée, our initial results for University Lille 1 suggest that practicing literacy through this encouragement device results in greater chances for students to complete the first year of university, or at least one of the semesters (Table 11b). In fact, encouraging first-year students to practice literacy during the first semester increases their probability of completing the first semester by 10 percentage points (at a 5.2 percent level). Moreover, improving literacy test scores by one additional point raises the probability of students completing their first semester by 6.5 percentage points.

Fourth, when we focus on different subpopulations (Tables 12 to 14), we see that the encouragement device does not provide the same level of benefit to all students.¹⁵ In particular, as with UPEM, our findings for the whole population of first-year students are mainly similar to those we get for male students (Tables 12a and 12b). Then, the encouragement device showed an increased benefit on students whose first language is French (Tables 13a and 13b), as was the case at UPEM. Finally, the experiment device appears to provide more benefit to students who hold a baccalaureate without honors (Tables 14a to 14c).

Fifth, for some disciplines, we observe a negative impact of the encouragement device, again for female students who demonstrate higher initial literacy skills than male students (Table 12b). This result suggests a lock-in effect: their time is diverted from other disciplines, although their literacy skills do not require intervention. We observe the same qualitative results as seen at UPEM.

Finally, these findings may be applied to help certain specific populations of students to complete their first year at university (or at least one of the semesters), in particular those students who initially have lower literacy skill levels.

As reported in Tables A12 to A14 in the appendix, encouraging students to practice literacy increases the probability that they will complete the first term of the academic year, by about 13 percentage points for male students, and by about 14 percentage points for university students who hold a baccalaureate without honors. Further, measurable improvements in literacy skills show similar benefits for these kind of students. In particular, a difference of one additional point in the measured improvement in scores on the literacy test increases the probability of completing the first term at university: by 9 percentage points for male students, by 10 percentage points for students who hold a baccalaureate without honors, and by 13 percentage points for those who hold a baccalaureate with minimal passing score (50%-60%), and by 8 percentage points for students whose native language is not French. ¹⁶

As with University Paris-Est Marne-La-Vallée, the implementation of the encouragement device may have contributed to raising academic achievement levels for first-year students at University Lille 1.

26

¹⁵ Like at UPEM, the initial literacy test score is on average lower among male students than among female students. It is higher for students who hold a baccalaureate with merit, honors or distinction than for other students. Finally, it is higher for students whose native language is French than for other students. Corresponding tables are available on request.

¹⁶ For female students however, we observe some lock-in effects: the probability of completing the second semester of the first year of university decreases as a consequence of the introduction of the encouragement device.

Table 12a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Managemer sample: male students.

sumpte:	mette stituetti	D•										
Discipline	Introduction to economics (GPA)	Introduction to economics (CA)	Introduction to management	Defining career objectives	Statistics (Semester 1)	English language (Semester 1) (CA)	Introduction to Macroeconomics	General accounting (GPA)	General accounting (FE)	Mathematics (F)	Mathematics (Exam 1)	Mathematics (Exam 2)
Model							Effect of encourages	ment to literacy lea	arning (Intention To treat	i, ITT)		
Without	0.697	0.987*	1.460**	1.596***	0.874	1.118*	1.205	0.680	0.660	0.963	0.482	1.142*
baseline	(0.454)	(0.517)	(0.633)	(0.403)	(0.599)	(0.575)	(0.749)	(0.574)	(0.576)	(0.626)	(0.651)	(0.607)
variables												
With	0.754*	0.981**	1.573**	1.669***	1.198**	3.174**	1.386*	0.943*	0.920	1.492***	0.945*	1.616***
baseline	(0.429)	(0.488)	(0.653)	(0.421)	(0.558)	(1.730)	(0.745)	(0.559)	(0.577)	(0.549)	(0.546)	(0.521)
variables												
Model						Eff	ect of varying literacy	test scores (Local	l Average Treatment Effe	ect, LATE)		
Without	0.512	0.727*	0.971**	1.201***	0.636	2.118**	0.922*	0.523	0.479	0.726	0.372	0.824*
baseline	(0.323)	(0.383)	(0.439)	(0.405)	(0.408	(0.901)	(0.555)	(0.415)	(0.399)	(0.468)	(0.490)	(0.433)
variables												
With	0.544*	0.702**	1.063**	1.224***	0.864**	2.089**	1.057**	0.708*	0.668*	1.131***	0.744*	1.165***
baseline	(0.293)	(0.346)	(0.464)	(0.406)	(0.363)	(0.860)	(0.534)	(0.403)	(0.368)	(0.425)	(0.413)	(0.365)
variables												` '

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 207 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student; the student; the student; the student; kind of baccalaureate (reference= Economics parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.460 point the score to the final exam in introduction to management for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 10 percent level, an increase of 1 point in the difference between to 0.922 point in the score in mathematics for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 12b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management sample: female students.

Discipli ne	Methodol	Defining	Financial mathematics (F)	English language (Semester 1)	English language (Semester 1)	General (CDA)	General (CA)	General (FF)	Mathematics (GPA)	Mathematics (Exam 1)	Mathematics (Exam 2)
ne	ogy (Sem.1)	career objectives	mathematics (1)	(CA)	(FE)	accounting (GPA)	accounting (CA)	accounting (FE)	(61.1)	(Balli 1)	(2)4411 2)
Model						Effect of er	ncouragement to literacy le	earning (Intention To treat, I	TT)		
Without	-1.222*	1.416***	-1.051	-1.020	-1.348	-1.839**	-2.001***	-1.633**	-1.511*	-2.345**	-1.882*
baseline	(0.644)	(0.475)	(0.749)	(1.624)	(1.847)	(0.722)	(0.688)	(0.791)	(0.839)	(0.900)	(0.841)
variables											
With	-1.486**	1.151***	-1.451*	-1.455**	-2.410*	-1.990***	-2.142***	-1.638**	-1.729***	-2.488***	-1.983***
baseline	(0.608)	(0.470)	(0.746)	(0.676)	(1.344)	(0.687)	(0.686)	(0.712)	(0.646)	(0.664)	(0.626)
variables											
Model						Effect of varyin	g literacy test scores (Loca	al Average Treatment Effect	, LATE)		
Without	-0.703	0.769**	-0.597	-0.564	-0.733	-0.998*	-1.034*	-0.898	-0.783	-1.177*	-0.994*
baseline	(0.475)	(0.299)	(0.540)	(0.972)	(1.082)	(0.590)	(0.559)	(0.600)	(0.541)	(0.624)	(0.582)
variables											
With	-1.003*	0.719**	-0.758	-0.916	-1.474	-1.162*	-1.189*	-0.965	-1.022*	-1.336**	-1.146*
baseline	(0.607)	(0.355)	(0.692)	(1.023)	(1.019)	(0.643)	(0.626)	(0.601)	(0.559)	(0.548)	(0.559)
variables											

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 116 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student; the student; kind of baccalaureate (reference= Econom within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice decreases by 1.455 point the score to the continuous assessment in English for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 10 percent level, an increase of 1 point in the difference between the sec

of 1.162 point in the score to the GPA in general accounting for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 13a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and

Management at university Lille 1. Considered sample: students whose country of origin is France.

Discipline	Introduction to economics (GPA)	Defining career objectives	Economic history	English language (Semester 1, CA)	Mathematics(CA)	Big contemporary economic issues
Model		Effec	t of encouragement to literacy	learning (Intention To treat,	ITT)	
Without	1.057**	1.635***	1.162	2.856**	2.201**	1.164*
baseline variables	(0.503)	(0.366)	(0.716)	(1.199)	(0.941)	(0.606)
With	0.973**	1.617***	1.214*	2.582**	2.973***	0.784
baseline variables	(0.503)	(0.362)	(0.643)	(1.123)	(0.861)	(0.596)
Model		Effect of	varying literacy test scores (L	ocal Average Treatment Effec	t, LATE)	
Without	0.595**	0.904***	0.638*	1.452**	1.397**	0.645*
baseline variables	(0.263)	(0.246)	(0.377)	(0.565)	(0.618)	(0.344)
With	0.559**	0.896***	0.673**	1.319**	1.974***	0.441
baseline variables	(0.248)	(0.243)	(0.339)	(0.529)	(0.663)	(0.334)

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 214 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 0.595 point the score to the GPA in introduction to economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.638 point in the score in economic history for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 13b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Lille 1. *Considered sample: students whose country of origin is NOT France.*

Discipline	Introduction to National accounting	Defining career objectives	Economic history	Functional analysis of organizations	Mathematics (Exam 1)	English language (Semester 2, GPA)	English language (Semester 2, FE)
Model			Effe	ect of encouragement to li	teracy learning (Intention To t	reat, ITT)	
Without baseline variables	-1.128*	1.593***	1.032	-1.625*	-1.587*	-1.615*	-3.120*
	(0.594)	(0.579)	(0.982)	(0.916)	(0.954)	(0.861)	(1.666)
With baseline variables	-0.724	1.730***	1.521*	-1.501*	-1.356	-1.454*	-3.136**
	(0.552)	(0.621)	(0.918)	(0.868)	0.818	(0.835)	(1.543)
Model			Effect of	f varying literacy test scor	es (Local Average Treatment	Effect, LATE)	
Without baseline variables	-0.829	1.181**	0.768	-1.223	-1.329	-1.231	-2.774
	(0.621)	(0.571)	(0.700)	(0.919)	(1.108)	(0.935)	(2.215)
With baseline variables	-0.443	1.159**	0.970*	-1.040	-1.008	-0.992	-2.425
	(0.378)	(0.509)	(0.573)	(0.741)	(0.743)	(0.723)	(1.623)

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 109 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 10 percent level, encouraging to literacy practice decreases by 1.128 point the score to introduction to national accounting for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 1.181 point in the score in "Defining career objectives" for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 14a. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management in university Lille 1. Considered sample: students who hold a baccalaureate with merit, honors or distinction.

Note	Introduction to national accounting	Defining career objectives	English language (Semester 1, FE)	Mathematics (Exam 1)	Mathematics (CC)	English language (Semester 2, GPA)	English language (Semester 2, CA)	English language (Semester 2, FE)
Model	decoditing		Effect of encouragen	nent to literacy learning (In	ntention To treat, IT	Γ)		
Without baseline	-0.897*	1.031**	-2.360*	-1.488*	2.455**	-1.806**	-3.563**	-3.445**
variables	(0.457)	(0.424)	(1.421)	(0.897)	(1.134)	(0.775)	(1.489)	(1.354)
With baseline	-0.621	1.031**	-1.298	-0.817	3.556***	-1.343*	-3.202**	-2.737**
variables	(0.432)	(0.431)	(1.300)	(0.763)	(1.095)	(0.771)	(1.532)	(1.211)
Model			Effect of varying literacy	test scores (Local Average	Treatment Effect, I	LATE)		
Without baseline	-0,492	0,552**	-1,240	-0,792	1,492**	-0,972*	-1,975*	-1,885*
variables	(0,311)	(0,231)	(0,869)	(0,563)	(0,741)	(0,562)	(1,142)	(0,998)
With baseline	-0,305	0,493***	-0,626	-0,399	2,012**	-0,670	-1,665	-1,398*
variables	(0,231)	(0,201)	(0,656)	(0,401)	(0,725)	(0,458)	(1,021)	(0,767)

Source: randomized experiment implemented in University Lille 1 (Northern France: 2013-2014).

Field: 136 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention fo treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student; the student; the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively *** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 1 percent level, encouraging to literacy practice increases by 2.455 point the score to the continuous assessment mathematics for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.493 point in the score in "Defining career objectives" for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 14b. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management in university Lille 1. *Considered sample: students who hold a baccalaureate without merit, honors or distinction.*

Note	Introduction	Introduction	Defining career	Statistics	Economic	English language	Mathematics(CA)	Tutorials in economics and
	to economics (GPA)	to management	objectives		history	(Semester, 1, CA)		computer science
Model			Effe	ect of encouragement	to literacy learning (Intenti	on To treat, ITT)		
Without baseline	1.268**	1.643**	1.963***	0.742	1.246*	3.110**	1.702*	0.986*
variables	(0.492)	(0.679)	(0.433)	(0.537)	(0.710)	(1.297)	(0.917)	(0.542)
With baseline	1.064**	1.590**	1.824***	0.881*	1.300*	2.767**	1.796**	0.906*
variables	(0.475)	(0.691)	(0.438)	(0.510)	(0.676)	(1.286)	(0.899)	(0.495)
Model			Effect of	f varying literacy test	scores (Local Average Tre	atment Effect, LATE)		
Without baseline	0.917**	1.108**	1.428***	0.537	0.869*	1.942**	1.349	0.745**
variables	(0.370)	(0.470)	(0.471)	(0.365)	(0.497)	(0.861)	(0.822)	(0.376)
With baseline	0.793**	1.094**	1.365***	0.657*	0.938*	1.763**	1.608*	0.696**
variables	(0.351)	(0.473)	(0.454)	(0.345)	(0.490)	(0.849)	(0.962)	(0.342)

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 187 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 1.590 point the score to the final exam in introduction to management for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0.793 point in the score in introduction to economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table 14c. Evaluation of the impact of the encouragement to literacy learning, or of varying literacy level on academic performance for first-year university students in Economic and Management at university Lille 1. Considered sample: students who got a baccalguréat with pass 50%-60%

Note	Introduction to economics (GPA)	Introduction to economics (CA)	Introduction to management	Defining career objectives	Statistics	Economic history	English language (Semester 1, CA)	General accounting (CA)	Mathematics(GPA)	Mathematics(CA)	Big contemporary economic issues	Tutorials in economics and computer science
Model					Effect of end	couragement to	literacy learning (Intention	To treat, ITT)				
Without	1.490**	1.015	1.455*	2.384***	0.840	0.940	3.355*	-1.422*	0.620	1.970	1.610**	1.139*
baseline	(0.571)	(0.676)	(0.851)	(0.593)	(0.724)	(0.927)	(1.744)	(0.745)	(0.722)	(1.188)	(0.757)	(0.667)
variables												
With	1.380***	1.015**	1.373	2.153***	1.269*	1.107	3.330*	-1.149	1.064*	2.436**	1.356*	1.188*
baseline	(0.545)	(0.597)	(0.905)	(0.614)	(0.691)	(0.865)	(1.820)	(0.706)	(0.568)	(1.082)	(0.818)	(0.633)
variables												
Model				I	Effect of varying	literacy test sc	ores (Local Average Treatm	nent Effect, LATE)				
Without	2.064	1.388	1.776	3.216*	1.164	1.277	3.859	-3.182	1.157	4.155	3.903	1.643
baseline	(1.290)	(1.144)	(1.238)	(1.927)	(0.986)	(1.345)	(2.577)	(3.817)	(1.585)	(4.774)	(4.409)	(1.138)
variables												
With	1.673*	1.488	1.508	2.601*	1.539*	1.380	3.300	-2.081	1.627	4.888	2.302	1.456*
baseline	(0.973)	(0.961)	(1.062)	(1.473)	(0.857)	(1.177)	(2.078)	(2.196)	(1.362)	(4.994)	(2.043)	(0.869)
variables												

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 118 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 1 percent level, encouraging to literacy practice increases by 1.380 point the score to the GPA in introduction to economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France). (b) At a 10 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 2.601 point in the score in "defining career objectives" for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

6. Conclusion

We have analyzed the impacts of encouraging and improving literacy skills on achieving academic success. We have implemented two randomized experiments in parallel with first-year students enrolled in Economics and Management at two universities: Paris-Est Marne-La-Vallée (2011-2014) and Lille 1 (2013-2014). The methodology of these experiments consisted in encouraging half of the student population in question to take advantage of an innovative pedagogical tool called *Projet Voltaire* for practicing literacy skills. The other half of our population was not encouraged at all.

Our results demonstrate that improving literacy skills can significantly increase academic performance in several disciplines, not only in language-related fields, but even more so in scientific ones. Depending on the discipline, scores can increase by 0.5 to 1.0-1.5 points on average. As a consequence, the probability of students completing the first year of university (or at least one of the semesters) also increases.

We also show that practicing literacy skills provides even more benefit to students who initially demonstrated low literacy skills, such as male students who earned their baccalaureate without honors or students whose native language is not French.

With these results, we have provided concrete evidence that higher literacy skills allow students to improve their academic performance during the first year of university, which is the crucial year for determining future prospects and, in the French context, the most competitive year. Our results underline the importance of public efforts to support literacy programs at the university level, and we further suggest that severe problems in basic literacy skills should be considered a crucial factor contributing to academic failure. Consequently, supporting and reinforcing literacy skills at the start of university programs is a key strategic variable that will prove essential to improving the overall success of programs, as well as the results of individual students throughout the path towards post-graduate studies and finally, to increase the chances of life-long success for all students.

References

Angrist J.D. and Imbens G. W. (1995), "Two-stage least squares estimation of average causal effects in models with variable treatment intensity", *Journal of the American Statistical Association*, Vol. 90, No. 430, pp. 431-442.

Angrist J.D., Graddy K. and Imbens G. W. (1996), "Interpretation of instrumental variables estimators in simultaneous equations models with an application to the demand for fish", *Review of Economic Studies*, Vol. 67, No. 3, pp. 499-527.

Angrist J.D., Imbens G. W. and Rubin D. B (1996), "Identification of causal effects using instrumental variables", *Journal of the American Statistical Association*, Vol. 91, No. 434, pp. 444-455.

Calmant J., Hallier P. (2008), "Etre diplômé de l'enseignement supérieur, un atout pour entrer dans la vie active", *Bref*, No. 253, Céreq.

Dearden L., Machin S. and Vignoles A. (2009), "Economics of education research: a review and future prospects", *Oxford Review of Education*, Vol. 35, No. 5, pp. 617–632.

Delgadova E. (2015), "Reading Literacy as One of the Most Significant Academic Competencies for the University Students", *Procedia - Social and Behavioral Sciences*, Vol. 178, pp. 48–53.

Duflo E., Glennerster R. and Kremer M. (2007), "Using randomization in development economics: a toolkit", *Handbook of Development Economics*, Vol. 4, Ch. 61, pp. 3895-3962.

Heckman J. J. and Vytlacil E. J. (2007), "Econometric evaluation of social programs, part I: causal models, structural models and econometric policy evaluation", in: J. Heckman & E. Leamer (Eds), *Handbook of econometrics*, Vol. 6B, pp. 4779–4874.

Heckman J. J. and Rubinstein Y. (2001), "The importance of noncognitive skills: lessons from the GED Testing Program", *American Economic Review*, Vol. 91, No. 2, pp. 145–149.

Imbens G. W. and Angrist J. D. (1994), "Identification and estimation of local average treatment effects", *Econometrica*, Vol. 62, No. 2, pp. 467-475.

Imbens, G. W. and Rubin D. B. (1997), "Estimating outcome distributions for compliers in instrumental variables models", *Review of Economic Studies*, Vol. 64, pp. 555-574.

Kaushik B. and Foster J.E. (1998), "On Measuring Literacy", *Economic Journal*, Vol. 108, No. 451, pp. 1733-1749.

Machin S. and McNally S. (2008), "The Literacy hour", *Journal of Public Economics*, Vol. 92, No. 92, Issues 5-6, pp. 1441–1462.

Mendez I. (2015), "The effect of the intergenerational transmission of noncognitive skills on student performance", *Economics of Education Review*, Vol. 46, pp. 78–97.

Murdoch J., Kamanzi P.C. and Doray P. (2011), "The influence of PISA scores, schooling and social factors on pathways to and within higher education in Canada", *Irish Educational Studies*, Vol. 30, No. 2, pp. 215-235.

OCDE (2000), "Literacy in the Information Age", *Report*.

Reardon S. F., Valentino R. A. and Kenneth A. (2012), "Patterns of Literacy among U.S. Students", *The Future of Children*, Vol. 22, No. 2, Literacy Challenges for the Twenty-First Century, pp. 17-37, Princeton University.

Rivkin S., Hanushek E. and Kain, J. (2005), "Teachers, schools, and academic achievement", *Econometrica*, vol. 73, p. 417–458.

Rubin D. B. (1974), « Estimating causal effects of treatments in randomized and non-randomized studies », *Journal of Educational Psychology*, vol. 66, pp. 688-701.

Appendices.

Table A1. Correlations between time to practice literacy and varying the literacy test score (difference between the second and the first literacy test scores), for first-year university students in Economics and Management, at university Paris-Est Marne-La-Vallée, considering different econometric specifications.

Time to literacy practice indicator	Indicator 1	Indicator 1	Indicator 1	Indicator 2	Indicator 2	Indicator 2
(specification)	(1)	(2)	(3)	(1)	(2)	(3)
/ Explanatory variables						
Time to literacy practice: indicator 1	0.012***	0.012***	0.011***			
	(0.001)	(0.001)	(0.001)			
Time to literacy practice: indicator 2				0.010***	0.010***	0.009***
				(0.001)	(0.001)	(0.001)
First literacy test score		0.096**	0.032		0.057	0.002
		(0.041)	(0.046)		(0.039)	(0.043)
Age			0.003			0.002
			(0.041)			(0.040)
Gender (Men vs. women)			-0.752***			-0.644***
			(0.252)			(0.240)
Scholarship student			0.068			0.026
•			(0.228)			(0.219)
Baccalaureate S			0.305			0.261
			(0.253)			(0.241)
Baccalaureate STG			-0.722*			-0.707*
			(0.429)			(0.414)
Other baccalaureate			-0.389			-0.383
			(0.442)			(0.415)
Intercept	1.954***	1.323***	2.152**	1.721***	1.351***	2.103**
•	(0.148)	(0.315)	(0.885)	(0.137)	(0.289)	(0.835)
Observations	528	528	528	528	528	528
R2	0.281	0.287	0.307	0.345	0.347	0.363
F	92.31	50.04	18.11	144.6	78.43	24.82

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year university students in Economics and Management stream, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as the two literacy test score.

Notes: correlation with or without control variables (OLS estimator). Time to practice literacy: indicator 1 = indicator provided by platform *Projet Voltaire*; indicator 2 = overall time spent using platform *Projet Voltaire* - duration of the 1st literacy evaluation - duration of the 2nd literacy evaluation. Explanatory variables: score to the first literacy test; first-year university student age; student gender (reference=female); scholarship student; kind of baccalaureate (reference= baccalaureate ES, Economics and Social Science stream). Robust standard errors. **** (respectively ** or *) stands for significance at a 1% (respectively a 5% or 10%).

Reading: at a 1 percent level, and whatever the considered econometric specification, a rise of 83 minutes in the time to literacy practice is associated to a 1 point more increase in the variation of the literacy test score, for first-year university students in Economics and Management at University Paris-Est Marne-La-Vallée.

Table A2. Effect of time to practice literacy on varying the literacy test score (difference between the second and the first literacy test scores), for first-year university students in Economics and Management, at university Paris-Est Marne-La-Vallée, considering different econometric specifications.

Time to literacy practice indicator (specification)	Indicator 1	Indicator 1	Indicator 1	Indicator 2	Indicator 2	Indicator 2
/ Explanatory variables	(1)	(2)	(3)	(1)	(2)	(3)
Time to literacy practice: indicator 1	0.021***	0.021***	0.021***			
	(0.004)	(0.004)	(0.004)			
Time to literacy practice: indicator 2				0.015***	0,015***	0.015***
J 1				(0,003)	(0.003)	(0.003)
Score to the first literacy test		0.110**	0.095*		0,045	0,030
·		(0.043)	(0.050)		(0,043)	(0,045)
Age			0.002			0,000
			(0.039)			(0,037)
Gender (Men vs. women)			-0.094			-0,051
			(0.396)			(0,380)
Scholarship student			-0.123			-0,155
			(0.264)			(0,249)
Baccalaureate S			0.170			0,124
			(0.280)			(0,265)
Baccalaureate STG			-0.434			-0,465
			(0.514)			(0,473)
Other baccalaureate			0.156			0,051
			(0.528)			(0,472)
Intercept	1.029**	0.331	0.452	0,870**	0,583	0,732
	(0.432)	(0.513)	(1.033)	(0,439)	(0,455)	(0,971)
Observations	526	526	526	526	526	526
R2	0.118	0.132	0.127	0,225	0,228	0,223
F	24.39	16.26	10.45	27,48	17,88	11,76

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014). Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as the two literacy test score.

Notes: effect of time to literacy practice (instrumental variable estimates using a Wald estimator instrumenting time to literacy practice using the encouragement dummy. Time to practice literacy: indicator 1 = indicator provided by platform *Projet Voltaire*; indicator 2 = overall time spent using platform *Projet Voltaire* - duration of the 1st literacy evaluation - duration of the 2nd literacy evaluation. Explanatory variables: score to the first literacy test; age of the first-year university student; gender of the student (reference=female); scholarship student; kind of baccalaureate (reference= baccalaureate ES, Economics and Social Science). Robust standard errors. *** (respectively ** or *) stands for significance at a 1% (respectively at a 5% or 10%) level.

Reading: at a 1 percent level, and considering the first indicator measuring the student time to literacy practice, an increase of 50 minutes in the time to literacy practice induces an increase in the variation of the 2 literacy tests scores for first-year university students in Economics and Management, at university Paris-Est Marne-La-Vallée.

34

Table A3. Effect of encouragement to literacy learning on academic performance of first-year university students at university Paris-Est Marne-La-Vallée. Detailed results.

	Overall GPA	First semester	GPA of Module 1	Introduction	Introduction	Introduction	Introduction	Introduction	Introduction
Variables / Grades	in first-year BA	GPA	(Semester 1)	to Economics(GPA)	to Economics (CA)	to Economics (FE)	to Management(GPA)	to Management(CA)	to Management (Fl
Encouragement	0.091	0.084	0.256	0.302	-0.293	0.708**	0.176	0.642***	-0.470
	(0.202)	(0.204)	(0.238)	(0.291)	(0.313)	(0.335)	(0.237)	(0.215)	(0.319)
Intercept	10.341***	10.272***	9.624***	9.191***	11.467***	7.292***	10.024***	11.438***	8.853***
•	(0.150)	(0.154)	(0.180)	(0.215)	(0.229)	(0.250)	(0.186)	(0.169)	(0.249)
Observations	437	492	516	521	517	512	519	517	518
R2	0.000	0.000	0.002	0.002	0.002	0.009	0.001	0.017	0.004
F	0.203	0.171	1.159	1.076	0.874	4.457	0.551	8.875	2.173

Variables / Grades	GPA of Module 2 (Semester 1)	Mathematics (GPA)	Mathematics (CA)	Mathematics (FE)	English language (GPA)	National Accounting (FE)	GPA of Module 3 (Semester 1)	Methodology (GPA)	Methodology (CA)	Methodology (FE)	Dissertation
Encouragement	0.187	0.533	-0.163	1.067**	0.061	-0.286	-0.150	0.125	0.045	-0.001	-0.213
Intercept	(0.275) 9.695***	(0.406) 9.031***	(0.405) 10.601***	(0.443) 7.764***	(0.237) 11.574***	(0.301) 8.739***	(0.207) 11.131***	(0.206) 10.245***	(0.199) 12.262***	(0.275) 8.538***	(0.195) 12.459***
	(0.211)	(0.304)	(0.306)	(0.331)	(0.188)	(0.228)	(0.150)	(0.158)	(0.142)	(0.216)	(0.129)
Observations	504	514	507	512	517	516	475	522	516	510	522
R2	0.001	0.003	0.000	0.011	0.000	0.002	0.001	0.001	0.000	0.000	0.002
F	0.460	1.724	0.162	5.803	0.0662	0.900	0.525	0.368	0.0521	1.48e-05	1.197

Variables /Grades	Second semester GPA	GPA of Module 1 (Semester 2)	Introduction to Microeconomics (GPA)	Introduction to Microeconomics (CA)	Introduction to Microeconomics (FE)	Introduction to Macroeconomics (GPA)	Introduction to Macroeconomics (CA)	Introduction to Macroeconomics (FE)	Financial economics (GPA)	Financial economics (CA)	Financial economics (FE)
E	0.162	0.256	1.082***	0.838**	0.655	0.200	0.074	0.000	0.507	0.607	0.052
Encouragement	0.162	0.356			0.655	0.398	0.074	-0.089	0.587	0.607	0.052
	(0.220)	(0.323)	(0.406)	(0.392)	(0.413)	(0.357)	(0.325)	(0.348)	(0.385)	(0.379)	(0.369)
Intercept	10.057***	8.999***	7.502***	9.665***	7.143***	8.691***	11.547***	7.753***	9.178***	11.102***	9.051***
•	(0.161)	(0.247)	(0.308)	(0.289)	(0.315)	(0.279)	(0.254)	(0.264)	(0.302)	(0.303)	(0.277)
Observations	443	466	493	454	451	492	453	453	488	454	452
R2	0.001	0.003	0.014	0.010	0.006	0.003	0.000	0.000	0.005	0.006	0.000
F	0.542	1.215	7.114	4.570	2.519	1.239	0.0514	0.0651	2.323	2.556	0.0197

Variables / Grades	GPA of Module 2 (Semester 2)	General policy of the firm (GPA)	Entrepreneurship (CA)	Entrepreneurship (FE)	Statistics and computer science (GPA)	Statistics and computer science (CA)	Statistics and computer science (FE)	GPA of Module 3 (Semester 2)	Principles Of law	English language (Semester 2)
Encouragement	0.240	0.350	-0.085	-0.172	0.875**	0.390	0.448	0.131	-0.118	0.384*
Encouragement	(0.268)	(0.302)	(0.226)	(0.283)	(0.369)	(0.292)	(0.396)	(0.175)	(0.222)	(0.230)
Intercept	9.517***	9.604***	11.796***	9.313***	8.234***	11.515***	7.152***	11.227***	11.056***	11.457***
mercept	(0.204)	(0.246)	(0.170)	(0.217)	(0.280)	(0.223)	(0.288)	(0.125)	(0.169)	(0.165)
Observations	469	491	462	454	489	439	455	447	454	449
R2	0.002	0.003	0.000	0.001	0.012	0.004	0.003	0.001	0.001	0.006
F	0.804	1.344	0.140	0.371	5.640	1.786	1.280	0.557	0.282	2.773

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).
Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator). Robust standard errors within parentheses. For a considered teaching, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** and *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: at a 5 percent level, encouraging to literacy learning increases by 1.082 points the score to the grade point average in introduction to microeconomics for first-year university students in Economics and Management at university Paris-Est Marne-La-Vallée (Paris region, France).

Table A4. Effect of increasing the literacy level on academic performance of first-year university students at university Paris-Est Marne-La-Vallée. *Detailed results*.

	Overall GPA	First semester	r GPA of Modu	le 1 Intro	oduction In	troduction	Introduction	Introduction	Introductio	n Intro	oduction
Variables / Grades	in first-year B	A GPA	(Semester 1)	to Econ	omics(GPA) to Eco	onomics (CA)	to Economics (FE)	to Management(GPA)	to Managemen	t(CA) to Mana	gement (FE)
Increase in literacy test score		0.068	0.203		0.246	-0.238	0.581**	0141	0.511***		0.404
	(0.196)	(0.159)	(0.178)		0.228)	(0.272)	(0.277)	(0181)	(0.173)		0.314)
Intercept	10.072***	10.190***	9.108***			2.072***	5.799***	9.666***	10.148***		395***
	(0.803)	(0.544)	(0.589)	((0.740)	(0.888)	(0.914)	(0.606)	(0.575)	(1	1.048)
Observations	437	492	516		521	517	512	519	517		518
R2	0.044	0.058	0.099		0.054	-0.135	-0.014	0.083	-0.028		0.304
F	0.0509	0.1800	1.290	1	1.160	0.760	4.370	0.600	8.640	1	1.650
	GPA of Module 2	Mathematics	Mathematics N	Lathematics	English language	National	GPA of Module	3 Methodology	Methodolo	gy Methodolog	y Dissertation
					8 8	Accounting				5,	,,
Variables / Grades	(Semester 1)	(GPA)	(CA)	(FE)	(GPA)	(FE)	(Semester 1)	(GPA)	(CA)	(FE)	
Increase in literacy test scores	0.153	0.428	-0.132	0.879**	0.051	-0.243	-0.117	0.103	0.037	-0.001	-0.176
mercuse in incrucy test scores	(0.216)	(0.306)	(0.337)	(0.343)	(0.195)	(0.276)	(0.168)	(0.165)	(0.158)	(0.233)	(0.178)
Intercept	9.293***	7.939***	10.942***	5.494***	11.445***	9.366***	11.420***	9.984***	12.168***		12.906***
	(0.731)	(1.012)	(1.126)	(1.135)	(0.654)	(0.911)	(0.543)	(0.543)	(0.514)	(0.786)	(0.559)
Observations	504	514	507	512	517	516	475	522	516	510	522
R2	0.084	0.120	-0.055	0.122	0.023	-0.156	-0.091	0.063	0.025	-0.001	-0.215
F	0.503	1.950	0.150	6.550	0.475	0.780	0.480	0.390	0.050	0.000	0.980
•	0.505	1.550	0.150	0.000	0.175	0.760	0.100	0.570	0.050	0.000	0.500
	Second semester (GPA of Module 1		Introduction to		Introduction	to Introduction to	Introduction to	Financial	Financial	Financial
Variables / Grades	GPA (Semester 2)		Microeconomic		Macroeconor			economics	economics	economics
			(GPA)	(CA)	(FE)	(GPA)	(CA)	(FE)	(GPA)	(CA)	(FE)
Increase in literacy test	0.167	0.385	1.032***	0.845**	0.690*	0.377	0.075	-0.093	0.558*	0.578*	0.055
scores											
	(0.213)	(0.324)	(0.376)	(0.402)	(0.420)	(0.315)	(0.325)	(0.372)	(0.337)	(0.352)	(0.389)
Intercept	9.575***	7.917***	4.799***	7.242***	5.170***	7.707***	* 11.332***	8.019***	7.695***	9.455***	8.892***
	(0.734)	(1.101)	(1.223)	(1.385)	(1.449)	(1.036)	(1.137)	(1.277)	(1.129)	(1.251)	(1.334)
Observations	443	466	493	454	451	492	453	453	488	454	452
R2	0.114	0.143	0.061	-0.026	0.070	0.137	0.031	-0.045	0.154	0.054	0.022
F	0.610	1.400	7.510	4.400	2.690	1.430	0.050	0.060	2.730	2.690	0.020
	GPA of Module 2	Entrepreneurship	Entrepreneurship	Policy	Descriptive star	tistics De	escriptive statistics	Descriptive statistics	GPA of Module 3	Principles	English languag
Variables / Grades	(Semester 2)	(GPA)	(CA)	of the firm	and computer s		l computer science	and computer science	(Semester 2)	Of law	(Semester 2)
variables / Grades				(FE)	(GPA)		(CA)	(FE)			
	0.249	0.329	-0.083	-0.187	0.840**		0.395	0.486	0.144	-0.134	0.430
	U.47	0.349			(0.335)		(0.294)	(0.407)	(0.189)	(0.260)	(0.272)
ncrease in literacy test scores	(0.261)	(0.267)	(0.227)								
•	(0.261) 8 828***	(0.267) 8 746***	(0.227) 12 029***	(0.326) 9 847***		:					, ,
•	(0.261) 8.828*** (0.876)	(0.267) 8.746*** (0.892)	(0.227) 12.029*** (0.773)	9.847*** (1.119)	6.037*** (1.091)	•	10.371*** (1.035)	5.760*** (1.378)	10.809*** (0.644)	(0.230) 11.442*** (0.889)	10.206*** (0.925)
ntercept	8.828*** (0.876)	8.746*** (0.892)	12.029*** (0.773)	9.847*** (1.119)	6.037*** (1.091)	•	10.371*** (1.035)	5.760*** (1.378)	10.809*** (0.644)	11.442*** (0.889)	10.206*** (0.925)
ncrease in literacy test scores ntercept Disservations	8.828*** (0.876)	8.746*** (0.892)	12.029*** (0.773)	9.847*** (1.119) 454	6.037*** (1.091)		10.371*** (1.035)	5.760*** (1.378) 455	10.809*** (0.644)	11.442*** (0.889)	10.206*** (0.925)
ntercept	8.828*** (0.876)	8.746*** (0.892)	12.029*** (0.773)	9.847*** (1.119)	6.037*** (1.091)	:	10.371*** (1.035)	5.760*** (1.378)	10.809*** (0.644)	11.442*** (0.889)	10.206*** (0.925)

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline survey is available, as well as the two literacy test score. Notes: effect of varying literacy test score (local average treatment effect; Wald estimator). Robust standard error within parentheses. *** (respectively ** and *) stands for significance at a 1% (respectively 5% and 10%) level.

Reading: at a 5 percent level, a rise of 1 point in the difference between the second and the first literacy test scores induces an increase of 0.879 point in the score of the final exam of Mathematics for first-year university students in Economics and Management at university Paris -Est Marne-La-Vallée.

Table A5. Effect of encouragement to literacy learning on academic performance of first-year university students at university Paris-Est Marne-La-Vallée.

	Overall GPA		GPA of Module		Introduction	Introduction	Introduction	Introduction		troduction
Variables / Scores	in first-year BA	GPA	(Semester 1)	to Economics(GPA	to Economics (CA)	to Economics (FE)	to Management(GPA)	to Management(CA) to Ma	nagement (FE)
Encouragement	0.203	0.115	0.250	0.290	-0.280	0.683**	0.190	0.627***		-0.448
2. neo urugemem	(0.177)	(0.171)	(0.212)	(0.262)	(0.292)	(0.313)	(0.213)	(0.206)		(0.294)
Score to the first literacy test	0.248***	0.280***	0.256***	0.295***	0.253***	0.339***	0.224***	0.160***		0.288***
	(0.038)	(0.038)	(0.047)	(0.059)	(0.064)	(0.068)	(0.046)	(0.043)		(0.062)
Age	0.009	0.013	-0.006	-0.000	-0.058	0.113*	-0.011	0.017		0.001
6.	(0.064)	(0.050)	(0.068)	(0.065)	(0.076)	(0.063)	(0.080)	(0.072)		(0.091)
Gender (Man vs. woman)	-0.919***	-0.916***	-0.686***	-0.277	-0.439	-0.250	-1.139***	-1.020***	-	1.321***
	(0.179)	(0.179)	(0.224)	(0.277)	(0.305)	(0.331)	(0.220)	(0.199)		(0.298)
Scholarship student	-0.358**	-0.301*	-0.328	-0.621**	-0.214	-0.884***	-0.065	-0.029		-0.105
Р	(0.179)	(0.174)	(0.218)	(0.273)	(0.308)	(0.324)	(0.213)	(0.204)		(0.298)
Baccalaureate S	0.744***	0.750***	0.500**	1.065***	1.271***	0.974***	-0.039	-0.125		0.246
Bucculation 5	(0.186)	(0.189)	(0.248)	(0.303)	(0.333)	(0.357)	(0.247)	(0.225)		(0.346)
Baccalaureate STG	-2.046***	-1.925***	-1.849***	-2.297***	-2.029***	-1.782***	-1.393***	-0.072		1.675***
Date and Die	(0.416)	(0.342)	(0.416)	(0.441)	(0.493)	(0.572)	(0.468)	(0.483)		(0.577)
Other baccalaureate	-0.613	-1.642***	-2.112***	-2.638***	-2.779***	-2.073**	-1.801***	-0.653		2.329***
omer succumureure	(0.788)	(0.616)	(0.670)	(0.814)	(0.945)	(0.836)	(0.600)	(0.581)		(0.709)
Intercept	8.960***	8.732***	8.647***	7.635***	11.093***	3.375**	9.626***	10.736***		7.870***
тегеорг	(1.237)	(1.003)	(1.338)	(1.334)	(1.476)	(1.322)	(1.562)	(1.409)		(1.800)
Observations	437	492	516	521	517	512	519	517		518
		0.308	0.195	0.193	0.152	0.144	0.172	0.109		0.150
R2	0.282	0.308 29.19	0.195 14.76	0.193 15.89	0.152 11.51	0.144 11.36	0.172 13.88	0.109 6.890		0.150 12.92
		0.308 29.19	0.195 14.76	0.193 15.89	0.152 11.51	0.144 11.36	0.172 13.88	0.109 6.890		0.150 12.92
R2	0.282	29.19	14.76	15.89	11.51	11.36		6.890	Methodology	12.92
R2	0.282 20.43	29.19	14.76	15.89 Mathematics Englis	11.51	11.36 GPA of M	13.88 Module 3 Methodology	6.890	Methodology (FE)	12.92
R2 F Variables / Scores	0.282 20.43 GPA of Module (Semester 1)	29.19 2 Mathematics (GPA)	Mathematics (CA)	Mathematics Englis (FE) (6	11.51 n language National A GPA) (F	11.36 Accounting GPA of N E) (Seme	13.88 Module 3 Methodology ster 1) (GPA)	Methodology M(CA)	(FE)	12.92 Dissertation
R2 F	0.282 20.43 GPA of Module (Semester 1) 0.191	29.19 2 Mathematics (GPA) 0.499	Mathematics (CA)	15.89 Mathematics Englis (FE) (0	11.51 I language National A GPA) (F	11.36 Accounting GPA of N (Seme	13.88 Module 3 Methodology (GPA) 80 0.129	6.890 Methodology (CA) 0.090	(FE) 0.033	Dissertation -0.179
R2 F Variables / Scores Encouragement	0.282 20.43 GPA of Module (Semester 1) 0.191 (0.230)	29.19 2 Mathematics (GPA) 0.499 (0.343)	14.76 Mathematics (CA) -0.156 (0.343)	15.89 Mathematics (FE) (1.036*** (0.387) (0.00000000000000000000000000000000000	11.51 I language National A (F) .052 -0.2 .206) (0.2	11.36 Accounting GPA of N (Seme 219 -0.1 81) (0.1	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183)	Methodology (CA) 0.090 (0.181)	(FE) 0.033 (0.259)	Dissertation -0.179 (0.168)
R2 F Variables / Scores Encouragement	0.282 20.43 GPA of Module (Semester 1) 0.191 (0.230) 0.313***	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291***	14.76 Mathematics (CA) -0.156 (0.343) 0.251***	15.89 Mathematics (FE) Englis (0.36*** (0.387) (0.335*** 0.4	11.51 h language National A GPA) (F 0.052 -0.2 2.206) (0.2 14*** 0.236	11.36 GPA of N (Seme 219 -0.181) (0.19*** 0.233	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) 9*** 0.272***	6.890 Methodology (CA) 0.090 (0.181) 0.275***	(FE) 0.033 (0.259) 0.318***	12.92 Dissertation -0.179 (0.168) 0.274***
R2 F Variables / Scores Encouragement Score to the first literacy test	0.282 20.43 GPA of Module (Semester 1) 0.191 (0.230) 0.313*** (0.049)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073)	15.89 Mathematics (FE) (0 1.036*** (0.387) ((0.35*** 0.4 (0.087) ((0.087) (0.087) ((0.087) (0.087) (0.087) (0.087) ((0.087) (0.087) (0.087) (0.087) ((0.087) (11.51 n language National A GPA) (F .052 -0.2 .206) (0.2 14*** 0.23 .043) (0.0	11.36 accounting GPA of N (Seme 219 -0.1 81) (0.1)*** 0.233 61) (0.0	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 92** 0.272*** 38) (0.041)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038)	(FE) 0.033 (0.259) 0.318*** (0.058)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037)
R2 F Variables / Scores Encouragement Score to the first literacy test	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037	15.89 Mathematics Englis (FE) (0 1.036*** (0.387) (0 0.335*** 0,4 (0.087) (0 0.166*	11.51 n language National A GPA) (F .052 -0.2 .206) (0.2 14*** 0.23 .043) (0.0 0.028 -0.0	11.36 GPA of N (Seme 219 -0.381) (0.1 (0.1 (0.14) (0.14) (0.14) (0.14) (0.14) (0.14) (0.14) (0.14) (0.14) (0.14)	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 13.88 0.272*** 0.272*** 0.08 -0.052	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012
R2 F Variables / Scores Encouragement Score to the first literacy test Age	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100)	15.89 Mathematics Englis (FE) (0.36*** (0.387) (0.335*** 0.4 (0.087) (0.166* (0.095) (0.995)	11.51 n language GPA) (F .052 -0206) (0.2 14*** 0.23 .043) (0.0 .028 -0.0 .057) (0.0	11.36 Accounting ED GPA of N (Seme 219 -0.181) (0.1)*** 0.23; (0.61) (0.0)449 -0.0,76) (0.0)	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) 0.272*** 338) (0.041) 108 -0.052 33) (0.044)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042)	-0.179 (0.168) 0.274*** (0.037) -0.012 (0.032)
R2 F Variables / Scores Encouragement Score to the first literacy test Age	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080***	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550***	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541***	15.89 Mathematics (FE) (1.036*** (0.387) ((0.087) (0.166* -(0.095) ((0.095) -1.429*** -0.000000000000000000000000000000000	11.51 n language PA (F) .052 -0206) (0.2 14*** 0.236 .043) (0.0 .028 -0.0 .057) (0.0 .367* -0.95	11.36 Accounting E) GPA of N (Seme 219 -0.181) (0.1)*** 0.23(61) (0.0)49 -0.6(76) (0.0)8*** -0.92	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 3*** 0.272*** 38) (0.041) 008 -0.052 33) (0.044) 7*** -0.922***	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654***	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134***	-0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884***
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman)	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357)	15.89 Mathematics (FE) Englis (0.36*** (0.387) (0.335*** 0.4 (0.087) (0.166* -(0.095) (0.71,229*** -0.0396) (0.396)	11.51 n language National A GPA) (F .052 -0.2 .206) (0.2 14*** 0.236 .043) (0.0 .028 -0.6 .057) (0.0 .367* -0.95 .208) (0.2	11.36 Accounting E) GPA of N (Seme 219 -0.1 81) (0.1 0.23 61) (0.0 0.49 -0.6 76) (0.0 8*** -0.92 83) (0.1	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) 38) (0.041) 008 -0.052 33) (0.044) 7*** -0.922*** 97) (0.187)	0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269)	-0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman)	0.282 20.43 GPA of Module (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505**	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674**	15.89 Mathematics (FE) 1.036*** (0.387) ((0.087) ((0.087) ((0.095) ((0.095) -1.429*** -0.066 -0.066 -0.066	11.51 n language National A 10.052 .206) 14*** 0.23 .043) 0.028 -0.0 .057) 0.067 .208) 0.028 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959 .0-0.959	11.36 accounting E) GPA of N (Seme 219 -0.181) (0.1 0.23) (61) (0.0 0.49 -0.0) (76) (0.0 8*** -0.92 83) (0.1 4*** -0.1	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) 8*** 0.272*** 38) (0.041) 108 -0.052 33) (0.044) 7*** -0.922*** 97) (0.187) 173 -0.236	0.090 (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370*	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338)	15.89 Mathematics (FE) (0 1.036*** (0.387) ((0.087) ((0.087) ((0.095) ((0.095) ((0.095) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.096) ((0.098) ((0	11.51 A language GPA National A GPA (F) 0.052	11.36 GPA of N (Seme 11.36 (Seme 11.	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 0.272*** 338) (0.041) 108 -0.052 333) (0.044) 7*** -0.922*** 97) (0.187) 73 -0.236 06) (0.187)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233) 1.636***	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569***	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611***	15.89 Mathematics (FE) 1.036*** (0.387) ((0.335*** 0.4 (0.097) ((0.166* -0.095) ((0.396) ((0.396) ((0.396) ((0.396) ((0.396) (0.396) ((0.388) (0.3617*** (0.098) (0.388) (0.3617*** (0.098) (0.388) (0.3617***	11.51 n language National A GPA) (F .052 -0.2 .206) (0.2 14*** 0.23 .043) (0.0 .028 -0.0 .057) (0.0 .367* -0.95 .208) (0.2 488** -0.78 .214) (0.2 .336 -0.87	11.36 Accounting ED GPA of N (Seme 219 -0.1 81) (0.1 9*** 0.23** 661) (0.0 9*** 0.92 9** 76) (0.0 8*** -0.92 83) (0.1 4*** -0.1 994) (0.2 3*** 0.1 1	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) 0.272*** 38) (0.041) 008 -0.052 33) (0.044) 7*** -0.922*** 97) (0.187) (73 -0.236 06) (0.187) 18 -0.016	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233) 1.636*** (0.253)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394)	15.89 Mathematics (FE) 1.036*** (0.387) ((0.087) ((0.087) ((0.087) ((0.087) (0.095) ((0.095) ((0.396) ((0.396) ((0.388) ((0.388) ((0.433	11.51 n language Pational A GPA) (F) .052	11.36 Accounting E) GPA of N (Seme 219 -0.181) (0.1 (0.2)	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 3*** 0.272*** 38) (0.041) 7*** -0.922*** 97) (0.187) 73 -0.236 06) (0.187) 18 -0.016 08) (0.201)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S	0.282 20.43 GPA of Module (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233) 1.636*** (0.253) -2.613***	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698***	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444***	15.89 Mathematics (FE) (1.036*** (0.387) (0.035*** 0.4 (0.087) (0.166* -1.429*** -0.066 -0.0388) (0.388) (0.388) (0.388) (0.388) (0.388) (0.388) (0.383) (3.617*** (0.433) (0.3726*** -1.	11.51 n language	11.36 Accounting E) GPA of N (Seme 219 -0.1 81) (0.1 9)*** 0.23; 61) (0.0 8*** -0.92 83) (0.1 4*** -0.1 994) (0.2 3*** 0.1 28) (0.2 3*** -0.7	13.88 Module 3 Methodology (GPA) 80 0.129 92) (0.183) *** 0.272*** 38) (0.041) 7*** -0.952 33) (0.044) 7*** -0.922*** 97) (0.187) 73 -0.236 06) (0.187) 18 -0.016 08) (0.201) 75* -1.515***	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610***	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814	-0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615***
Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.233) 1.636*** (0.253) -2.613*** (0.423)	29.19 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570)	15.89 Mathematics (FE) (0 1.036*** (0.387) ((0.387) (0.087) ((0.087) (0.166* -0.095) (0.166* -0.0966 (0.388) (0.388) (0.388) (0.388) (0.388) (0.3726*** -1. (0.662) (0.662)	11.51 A language GPA National A GPA	11.36 GPA of N (Seme 11.36 (Seme 11.	13.88	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578)
Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.233) 1.636*** (0.233) -2.613*** (0.423) -1.999**	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576) -1.867*	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570) -1.916	15.89 Mathematics (FE) (0 1.036*** (0.387) (0.335*** 0.4 (0.087) (0.166* -0.095) (0.166* -0.0366 -0.066 (0.388) (0.388) (0.388) (0.388) (0.3726*** -1. (0.662) (0.1435 -2.1435 -2.1	11.51 A language GPA National A	11.36 GPA of N (Seme 219 -0.1 81) (0.1 94) (0.6 94) (0.2 3*** -0.1 94) (0.2 3*** 0.1 28) (0.2 3*** 0.1 28) (0.2 3*** -0.7 19) (0.4 7*** -0.1	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 0.272*** 338) (0.041) 108 -0.052 333) (0.044) 7*** -0.922*** 97) (0.187) 73 -0.236 06) (0.187) 18 -0.016 08) (0.201) 75* -1.515*** 61) (0.355) 199 -1.606**	6.890 Methodology (CA) 0.090 (0.181) 0.275**** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452) -1.300**	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532) -1.134	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578) -0.737
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG Other baccalaureate	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233) 1.636*** (0.253) -2.613*** (0.423) -1.999** (0.776)	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576) -1.867* (1.110)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570) -1.916 (1.188)	15.89 Mathematics (FE) 1.036*** (0.387) ((0.387) ((0.087) ((0.087) (0.166* (-0.095) ((0.396) ((0.396) ((0.396) ((0.396) ((0.396) ((0.388) ((0.433) ((0.433) ((0.433) ((0.662) ((0.662) ((0.662) ((0.662) ((0.662) ((0.186	11.51 A language	11.36 Accounting E) GPA of N (Seme 219 -0.181) (0.1 0.23) (61) (0.0 0.49) -0.6 (76) (0.0 8*** -0.92 83) (0.1 4*** -0.194) (0.2 3*** 0.1 28) (0.2 3*** 0.1 19) (0.4 75** 0.9) (0.6 0.9)	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 18** 0.272*** 188 (0.041) 189 -0.052 189 (0.187) 173 -0.236 180 (0.187) 18 -0.016 180 (0.201) 175* -1.515*** 11.606** 146) (0.649)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452) -1.300** (0.637)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532) -1.134 (0.715)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578) -0.737 (0.551)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG Other baccalaureate	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.233) 1.636*** (0.233) -2.613*** (0.423) -1.999**	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576) -1.867*	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570) -1.916	15.89 Mathematics (FE) (0 1.036*** (0.387) (0 0.335*** 0.4 (0.087) (0 0.166* (0.095) (0 -1.429*** (0.396) (0 -0.066 (0.388) (0 0.388) (0 3.617*** (0.433) (0 -3.726*** -1. (0.662) (0 -1.435 -2. (1.186) (0 2.612 9.8	11.51 A language GPA National A	11.36 Accounting E) GPA of N (Seme 219 -0.181) (0.1 (0.0 (0.0 (0.1 (0.1 (0.1 (0.1 (0.1	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 3*** 0.272*** 38) (0.041) 908 -0.052 33) (0.044) 7*** -0.922*** 97) (0.187) 73 -0.236 06) (0.187) 18 -0.016 08) (0.201) 75* -1.515*** 61) (0.355) 99 -1.606* 46) (0.649) 2*** 10.233***	6.890 Methodology (CA) 0.090 (0.181) 0.275**** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452) -1.300**	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532) -1.134	-0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578) -0.737 (0.551)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG Other baccalaureate Intercept	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.233) 1.636*** (0.253) -2.613*** (0.423) -1.999** (0.776) 7.665*** (1.275)	29.19 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576) -1.867* (1.110) 5.812*** (1.798)	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570) -1.916 (1.188) 8.638*** (1.966)	15.89 Mathematics (FE) (0 1.036*** (0.387) ((0.335*** 0.4 (0.087) ((0.095) (0.166* -0.095) ((0.396) (0.396) ((0.396) (0.388) (0.388) (0.3617*** (0.433) ((0.433) (0.3726*** -1. (0.662) (1.1435 (-1.435 (1.186) ((1.907) (1.907) (1.1907) (1.1907) (1.1907)	11.51 Alanguage National A	11.36 GPA of N (Seme 11.36 (Seme 11.	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 0.272*** 338) (0.041) 108 -0.052 333) (0.044) 7*** -0.922*** 97) (0.187) 173 -0.236 06) (0.187) 18 -0.016 08) (0.201) 775* -1.515*** 61) (0.355) 1606** 46) (0.649) 22*** 10.233*** 71) (0.853)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452) -1.300** (0.637) 11.793*** (1.074)	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532) -1.134 (0.715) 8.072*** (0.887)	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578) -0.737 (0.551) 11.633*** (0.629)
R2 F Variables / Scores Encouragement Score to the first literacy test Age Gender (Man vs. woman) Scholarship student Baccalaureate S Baccalaureate STG Other baccalaureate	0.282 20.43 GPA of Module: (Semester 1) 0.191 (0.230) 0.313*** (0.049) 0.029 (0.064) -1.080*** (0.237) -0.505** (0.233) 1.636*** (0.253) -2.613*** (0.423) -1.999** (0.776) 7.665***	29.19 2 Mathematics (GPA) 0.499 (0.343) 0.291*** (0.075) 0.091 (0.090) -1.550*** (0.354) -0.494 (0.343) 3.569*** (0.386) -3.698*** (0.576) -1.867* (1.110) 5.812***	14.76 Mathematics (CA) -0.156 (0.343) 0.251*** (0.073) 0.037 (0.100) -1.541*** (0.357) -0.674** (0.338) 3.611*** (0.394) -3.444*** (0.570) -1.916 (1.188) 8.638***	15.89 Mathematics (FE) 1.036*** (0.387) ((0.335*** 0.4 (0.087) ((0.095) (0.166* -0.095) ((0.396) (0.396) ((0.396) (0.388) (0.388) ((0.388) (0.3726*** (0.433) ((0.662) (1.435	11.51 n language National A GPA) (F .052 -0.2 .206) (0.2 .14*** 0.236 .043) (0.6 .057) (0.6 .367* -0.95 .208) (0.2 488** -0.78 .214) (0.2 .336 -0.87 .210) (0.3 86*** -2.26 .393) (0.5 .14*** -2.34 .728) (0.7	11.36 GPA of NE (Seme Company Company	13.88 Module 3 Methodology (GPA) 180 0.129 92) (0.183) 0.272*** 338) (0.041) 108 -0.052 333) (0.044) 7*** -0.922*** 97) (0.187) 173 -0.236 06) (0.187) 18 -0.016 08) (0.201) 75* -1.515** 61) (0.355) 199 -1.606** 46) (0.649) 2*** 10.233*** 71) (0.853)	6.890 Methodology (CA) 0.090 (0.181) 0.275*** (0.038) -0.032 (0.056) -0.654*** (0.185) -0.370* (0.190) -0.226 (0.197) -1.610*** (0.452) -1.300** (0.637) 11.793***	(FE) 0.033 (0.259) 0.318*** (0.058) -0.048 (0.042) -1.134*** (0.269) -0.149 (0.264) 0.228 (0.278) -0.814 (0.532) -1.134 (0.715) 8.072***	12.92 Dissertation -0.179 (0.168) 0.274*** (0.037) -0.012 (0.032) -0.884*** (0.186) -0.113 (0.184) -0.221 (0.177) -1.615*** (0.578) -0.737 (0.551) 11.633***

Variables / Scores	Second semester GPA	GPA of Module 1 (Semester 2)	Introduction to Microeconomics (GPA)	Introduction to Microeconomics (CA)	Introduction to Microeconomics (FE)	Introduction to Macroeconomics (GPA)	Introduction to Macroeconomics (CA)	Introduction to Macroeconomics (FE)	Financial economics (GPA)	Financial economics (CA)	Financial economics (FE)
Encouragement	0.312	0.508*	1.048***	0.954***	0.828**	0.371	0.241	0.051	0.619*	0.728**	0.262
Encouragement	(0.199)									(0.360)	
Pine User		(0.298)	(0.367)	(0.366)	(0.396)	(0.321)	(0.307)	(0.334)	(0.353)	()	(0.356)
First literacy	0.246***	0.235***	0.282***	0.163**	0.271***	0.343***	0.198***	0.336***	0.258***	0.234***	0.230***
test score						(0.000)	(0.041)				
	(0.041)	(0.063)	(0.082)	(0.081)	(0.085)	(0.068)	(0.061)	(0.070)	(0.073)	(0.075)	(0.070)
Age	-0.023	-0.028	0.106	0.000	0.066	0.068	-0.035	-0.015	0.090	-0.002	-0.071
	(0.073)	(0.086)	(0.086)	(0.068)	(0.094)	(0.106)	(0.089)	(0.102)	(0.122)	(0.094)	(0.141)
Gender (Man vs.	-0.912***	-1.325***	-2.159***	-1.782***	-1.650***	-1.140***	-1.165***	-0.110	-1.609***	-1.557***	-0.720**
woman)											
	(0.201)	(0.302)	(0.373)	(0.368)	(0.400)	(0.314)	(0.303)	(0.334)	(0.354)	(0.365)	(0.364)
Scholarship student	-0.346*	-0.269	0.037	-0.069	-0.420	0.030	-0.079	-0.501	-0.444	-0.554	-0.823**
•	(0.203)	(0.305)	(0.373)	(0.371)	(0.401)	(0.320)	(0.319)	(0.333)	(0.353)	(0.368)	(0.361)
Baccalaureate S	0.696***	0.848***	1.881***	2.554***	1.047**	0.683*	0.069	1.133***	-0.222	-0.185	-0.349
	(0.209)	(0.317)	(0.405)	(0.395)	(0.429)	(0.349)	(0.342)	(0.352)	(0.373)	(0.392)	(0.377)
Baccalaureate STG	-2.157***	-3.036***	-2.965***	-2.412***	-2.846***	-3.378***	-2.838***	-2.339***	-3.462***	-2.286***	-2.815***
Daccana arcate 516	(0.438)	(0.605)	(0.666)	(0.816)	(0.782)	(0.623)	(0.617)	(0.653)	(0.659)	(0.664)	(0.798)
Other baccalaureate	-0.468	-1.391	-0.741	0.041	0.104	-2.974***	-2.795**	-1.232	-3.238**	-2.018	-2.103
Other baccalaureate	(0.908)	(1.219)	(1.289)	(1.484)	(1.490)	(1.071)	(1.200)	(1.002)	(1.295)	(1.316)	(1.378)
Intercent	9.282***	8.661***	4.577**	8.818***	4.885**	5.964***	11.651***	5.782***	7.291***	10.808***	9.748***
Intercept	(1.410)	(1.741)	(1.778)		(1.894)	(2.082)	(1.739)	(2.007)	(2.383)	(1.825)	
	(1.410)	(1.741)	(1.//6)	(1.451)	(1.694)	(2.062)	(1./39)	(2.007)	(2.363)	(1.625)	(2.721)
Observations	443	466	493	454	451	492	453	453	488	454	452
R2	0.240	0.181	0.216	0.183	0.134	0.209	0.133	0.137	0.172	0.117	0.100
F	17.34	12.06	17.60	12.50	8.806	14.70	7.687	10.46	11.82	7.449	6.285

Variables / Scores	GPA of Module 2 (Semester 2)	Entrepreneurship (GPA)	Entrepreneurship (CA)	Entrepreneurship (FE)	Statistics and computer science (GPA)	Statistics and computer science (CA)	Statistics and computer science (FE)	GPA of Module 3 (Semester 2)	Principles Of law t	English language (Semester 2)
Encouragement	0.289	0.294	-0.058	-0.067	0.807**	0.437	0.641*	0.236	0.014	0.477**
<u> </u>	(0.243)	(0.281)	(0.220)	(0.281)	(0.324)	(0.269)	(0.365)	(0.164)	(0.223)	(0.206)
First literacy test score	0.289***	0.234***	0.154***	0.201***	0.418***	0.269***	0.452***	0.204***	0.054	0.372***
	(0.051)	(0.059)	(0.048)	(0.063)	(0.071)	(0.052)	(0.078)	(0.034)	(0.045)	(0.046)
Age	0.008	0.125	0.053	-0.037	0.134	-0.041	-0.002	-0.106**	-0.170***	-0.062
	(0.089)	(0.098)	(0.111)	(0.069)	(0.103)	(0.093)	(0.125)	(0.043)	(0.045)	(0.074)
Gender (Man vs. woman)	-0.630**	-0.940***	-0.492**	-0.478*	-0.817**	-0.139	-0.668*	-0.491***	-0.237	-0.553***
	(0.244)	(0.256)	(0.216)	(0.280)	(0.325)	(0.271)	(0.375)	(0.173)	(0.228)	(0.209)
Scholarship student	-0.433*	-0.108	-0.723***	-0.406	-0.198	-0.533**	-0.620*	-0.046	0.053	-0.179
•	(0.239)	(0.264)	(0.222)	(0.273)	(0.318)	(0.259)	(0.369)	(0.166)	(0.218)	(0.214)
Baccalaureate S	1.062***	0.088	-0.139	0.087	1.677***	1.356***	1.873***	0.233	-0.179	0.486**
	(0.254)	(0.287)	(0.230)	(0.298)	(0.363)	(0.296)	(0.414)	(0.174)	(0.235)	(0.227)
Baccalaureate STG	-2.050***	-2.483***	-1.184**	-1.169**	-3.413***	-2.333***	-3.029***	-1.232***	-1.180**	-1.344**
	(0.429)	(0.636)	(0.490)	(0.460)	(0.571)	(0.538)	(0.610)	(0.445)	(0.542)	(0.598)
Other baccalaureate	-1.822*	-3.244***	-2.409***	-0.589	-2.059*	-1.731	-0.497	-0.415	-0.352	-0.860
	(1.050)	(1.115)	(0.898)	(1.319)	(1.113)	(1.058)	(1.286)	(0.516)	(0.632)	(0.587)
Intercept	7.790***	6.617***	10.499***	9.081***	3.386*	10.475***	4.275*	12.063***	14.015***	10.369***
	(1.727)	(1.898)	(2.096)	(1.373)	(2.044)	(1.775)	(2.429)	(0.839)	(0.892)	(1.466)
Observations	469	491	462	454	489	439	455	447	454	449
R2	0.211	0.175	0.111	0.064	0.257	0.195	0.205	0.164	0.045	0.227
F	16.89	9.223	5.827	4.543	23.55	13.97	19.91	10.92	2.759	14.82

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year university students in Economics and Management stream, entering device for the first year, for whom information from the baseline survey is available, as well as the two literacy test score. Notes: effect of encouragement to literacy learning (intention to treat). Robust standard errors within parentheses. For a considered teaching, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to FE and CA. Included baseline variables: score to the first literacy test, age of the university student; gender (reference=female); scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). *** (respectively ** and *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: at a 5 percent level, encouraging to literacy learning increases by 1.048 points the score to the grade point average in introduction to microeconomics for first-year university students in Economics and Management at university Paris-Est Marne-La-Vallée (Paris region, France).

Table A6. Effect of increasing the literacy level on academic performance of first-year university students at university Paris-Est Marne-La-Vallée. *Information provided by the administrative baseline survey are used are explanatory variables of the considered score.* Detailed results.

	Overall GPA	First semester	GPA of Module			troduction	Introduction		duction	Introductio		ntroduction
Variables / Grades	in first-year BA	GPA	(Semester 1)	to Economi	ics(GPA) to Eco	onomics (CA) to	o Economics (FE	to Manage	ement(GPA)	to Management	t(CA) to Ma	nagement (FE)
Increase in literacy test scores	0.186	0.088	0.196	0.23		-0.226	0.552**		.150	0.494***		-0.374
	(0.150)	(0.124)	(0.157)	(0.20		(0.248)	(0.253)		.160)	(0.167)		(0.273)
Score to the first literacy test	0.262***	0.285***	0.266***	0.305).242***	0.362***		32***	0.185***		0.272***
	(0.037)	(0.036)	(0.046)	(0.05		(0.068)	(0.067)		.045)	(0.046)		(0.069)
Age	0.016	0.012	-0.005	0.00		-0.057	0.112*		.011	0.016		-0.000
	(0.062)	(0.049)	(0.064)	(0.06		(0.081)	(0.059)		.078)	(0.070)		(0.097)
Gender (Men vs. women)	-0.608**	-0.787***	-0.397	0.07		-0.794	0.570		17***	-0.271		-1.879***
	(0.302)	(0.238)	(0.306)	(0.39		(0.494)	(0.508)		.311)	(0.332)		(0.510)
Scholarship student	-0.391**	-0.328*	-0.383*	-0.679		-0.148	-1.024***		.106	-0.161		-0.023
	(0.166)	(0.171)	(0.213)	(0.26		(0.330)	(0.327)		.209)	(0.213)		(0.336)
Baccalaureate S	0.654***	0.721***	0.428*	0.972		1.366***	0.737*		.094	-0.335		0.402
	(0.197)	(0.190)	(0.251)	(0.30		(0.369)	(0.382)		.255)	(0.257)		(0.399)
Baccalaureate STG	-1.875***	-1.855***	-1.619***	-2.037		2.271***	-1.190*		213**	0.440		-2.098***
	(0.413)	(0.350)	(0.450)	(0.50		(0.560)	(0.708)		.494)	(0.500)		(0.669)
Other Baccalaureate	-0.559	-1.567***	-1.910***	-2.413		3.011***	-1.616**		40***	-0.116		-2.694***
	(0.748)	(0.599)	(0.638)	(0.78		(1.034)	(0.794)		.581)	(0.565)		(0.816)
Intercept	8.081***	8.422***	7.926***	6.784	*** 1	1.904***	1.440	9.07	73***	8.987***		9.243***
	(1.395)	(1.084)	(1.441)	(1.49	96)	(1.819)	(1.590)	(1.	.664)	(1.561)		(2.188)
Observations	437	492	516	52	1	517	512	5	519	517		518
				0.0		0.051	0.112	0	.229	0.030		-0.068
R2	0.369	0.362	0.261	0.24	ł I	0.051		U.	.449			
	0.369 22.82	0.362 31.02	0.261 16.03	0.24 16.1		10.49	10.75		5.36	7.098		10.18
R2 F	22.82	31.02	16.03	16.1	15	10.49	10.75	15	5.36	7.098		10.18
F	22.82 GPA of Module 2	31.02 2 Mathematics	Mathematics	16.1	5 English languag	e National Acco	10.75 ounting GPA of	Module 3	5.36 Methodology	7.098 Methodology		10.18
	22.82	31.02	16.03	16.1	15	10.49	10.75 ounting GPA of	15	5.36	7.098	Methodology (FE)	
F Variables / Grades	GPA of Module 2 (Semester 1)	31.02 Mathematics (GPA)	Mathematics (CA)	Mathematics (FE)	English languag (GPA)	e National Acco	Dunting GPA of (Seme	Module 3 :	Methodology (GPA)	7.098 Methodology (CA)	(FE)	10.18 Dissertation
F Variables / Grades	GPA of Module (Semester 1) 0.152	31.02 2 Mathematics (GPA) 0.395	Mathematics (CA)	Mathematics (FE)	English languag (GPA)	e National Acco (FE)	Dunting GPA of (Seme	Module 3 Sester 1)	Methodology (GPA) 0.106	7.098 Methodology (CA) 0.072	(FE) 0.027	10.18 Dissertation -0.146
F Variables / Grades Increase in literacy test scores	22.82 GPA of Module 2 (Semester 1) 0.152 (0.174)	31.02 2 Mathematics (GPA) 0.395 (0.255)	16.03 Mathematics (CA) -0.124 (0.278)	Mathematics (FE) 0.836*** (0.298)	English languag (GPA) 0.042 (0.165)	10.49 e National Accc (FE) -0.182 (0.244)	Dunting GPA of (Seme	Module 3 :: ester 1) :: 140 :: 155)	5.36 Methodology (GPA) 0.106 (0.145)	7.098 Methodology (CA) 0.072 (0.139)	(FE) 0.027 (0.210)	10.18 Dissertation -0.146 (0.148)
F Variables / Grades Increase in literacy test scores	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320***	31.02 2 Mathematics (GPA) 0.395 (0.255) 0.310***	16.03 Mathematics (CA) -0.124 (0.278) 0.244***	Mathematics (FE) 0.836*** (0.298) 0.370***	English languag (GPA) 0.042 (0.165) 0.416***	e National Acco (FE) -0.182 (0.244) 0.222**	10.75 Dunting GPA of (Seme -0. (0. * 0.22	Module 3 (ster 1) 140 155) 9***	Methodology (GPA) 0.106 (0.145) 0.277***	7.098 Methodology (CA) 0.072 (0.139) 0.278***	(FE) 0.027 (0.210) 0.318***	10.18 Dissertation -0.146 (0.148) 0.268***
Variables / Grades Increase in literacy test scores Score to the first literacy test	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046)	31.02 2 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043)	10.49 e National According (FE) -0.182 (0.244) 0.222** (0.063)	Dunting GPA of (Seme Co.) -0. (0.) * 0.22	Module 3 ester 1) 140 155) 9*** 039)	Methodology (GPA) 0.106 (0.145) 0.277*** (0.041)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038)	(FE) 0.027 (0.210) 0.318*** (0.057)	-0.146 (0.148) 0.268*** (0.040)
Variables / Grades Increase in literacy test scores Score to the first literacy test	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029	31.02 2 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168**	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028	10.49 e National According (FE) -0.182	Dunting GPA of (Seme -0. * 0.22 (0.1 -0000000000.	Module 3 ester 1) 140 155) 9*** 039) 005	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032	0.027 (0.210) 0.318*** (0.057) -0.048	10.18 -0.146 (0.148) 0.268*** (0.040) -0.013
F Variables / Grades Increase in literacy test scores Score to the first literacy test Age	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086)	16.03 Mathematics (CA) -0.124 (0.278) 0.244**** (0.077) 0.037 (0.099)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057)	10.49 e National Acc (FE) -0.182 (0.244) (0.222** (0.063) -0.050 (0.076)	10.75 Dunting GPA of (Seme -0. (0. * 0.22 (0.) -0. (0.) (0.)	Module 3 seter 1) 140 155) 9*** 339) 005 334)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034)
F Variables / Grades Increase in literacy test scores Score to the first literacy test Age	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848**	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952*	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736***	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229**	Dunting GPA of (Seme * 0.22 (0.1	Module 3 seter 1) 140 155) 99** 309) 005 034) 18***	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763***	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544*	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094***	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108***
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women)	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323)	10.49 e National According (FE) -0.182 (0.244) 0.222*** (0.063) -0.050 (0.076) -1.229*** (0.447)	10.75 Dunting GPA of (Semi variable) -0. (0. * 0.22 (00001.14 (01.14	15 Module 3 sister 1) 140 155) 9*** 139) 005 1034) 18*** 333)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303)
F Variables / Grades	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547**	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601*	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635*	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499**	10.49 e National Accor (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742*	10.75 Dunting GPA of (Seme 1.1.1.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4.4	Module 3 ster 1) 140 155) 9*** 339) 005 334) 18** 8** 333)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390**	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217)	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310)	Dunting GPA of (Seme Control of C	Module 3 sester 1) 140 155) 9*** 039) 005 034) 8*** 333) 119 236)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185)	7.098 Methodology (CA) 0.072 (0.139) 0.278**** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190)	0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204)
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women)	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577***	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419***	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661***	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287***	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319	10.49 e National Accc (FE) -0.182 (0.244) 0.222*** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797*	Dunting GPA of (Seme Continuo	Module 3 sester 1) 140 155) 9*** 039) 005 034) 18*** 119 236) 183	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254	0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204) -0.163
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217)	10.49 e National Acco (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351)	# 10.75 Ounting GPA of (Seme 1.0	15 Module 3 seter 1) 140 155) 9*** 339) 005 134) 18*** 333) 119 2236) 183 2226)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204) -0.163 (0.201)
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476***	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292***	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560***	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811***	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143***	10.49 e National Acco (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462**	# 10.75 Ounting GPA of (Semi Part of Semi P	Module 3 sister 1) 140 155) 9*** 339) 005 334) 18*** 333) 119 236) 1883 226) 118*	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392***	7.098 Methodology (CA) 0.072 (0.139) 0.278**** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390*** (0.190) -0.254 (0.201) -1.526***	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.204) -0.163 (0.201) -1.784***
F Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.410*** (0.393) -3.292*** (0.604)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143*** (0.405)	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.347) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576)	Dunting GPA of (Seme Control of Seme Control o	Module 3 sester 1) 140	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204) -0.163 (0.201) -1.784*** (0.649)
F Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547*** (0.266) 1.577*** (0.260) -2.476*** (0.434) -1.861**	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.427) -2.026	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143*** (0.405) -1.977***	10.49 e National Accc (FE) -0.182 (0.244) 0.222*** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742** (0.310) -0.797* (0.351) -2.462** (0.576) -2.524**	### 10.75 Dunting GPA of (Seme 10.22	Module 3 sester 1) 140 155) 9*** 039) 005 034) 18*** 333) 119 236) 183 226) 184 503) 956	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502**	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239**	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG Other Baccalaureate	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320**** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434) -1.861** (0.761)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474 (1.057)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637) -2.026 (1.242)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717 (1.072)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143*** (0.405) -1.977*** (0.724)	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576) -2.524** (0.768)	### 10.75 Dunting GPA of (Semi Continuo Conti	Module 3 sester 1) 140 1555) 9*** 039) 005 034) 18*** 333) 1119 236) 188 226) 118* 503) 956 666)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502** (0.620)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239** (0.621)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111 (0.731)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879 (0.593)
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG Other Baccalaureate	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434) -1.861** (0.761) 7.118***	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474 (1.057) 4.368**	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637) -2.026 (1.242) 9.083***	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717 (1.072) -0.418	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) -1.143*** (0.405) -1.977*** (0.724) 9.707***	10.49 e National Acco (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576) -2.524** (0.768) 10.128**	### 10.75 Ounting GPA of (Semi Continuous	Module 3 sester 1) 140 155) 9*** 339) 005 334) 18*** 333) 119 2236) 188 226) 118* 503) 9956 666) 53***	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502** (0.620) 9.841***	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239** (0.4621) 11.548***	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111 (0.731) 7.979***	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879 (0.593) 12.172***
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320**** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434) -1.861** (0.761)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474 (1.057)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637) -2.026 (1.242)	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717 (1.072)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143*** (0.405) -1.977*** (0.724)	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576) -2.524** (0.768)	### 10.75 Ounting GPA of (Semi Continuous	Module 3 sester 1) 140 1555) 9*** 039) 005 034) 18*** 333) 1119 236) 188 226) 118* 503) 956 666)	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502** (0.620)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239** (0.621)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111 (0.731)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879
Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG Other Baccalaureate	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434) -1.861** (0.761) 7.118***	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474 (1.057) 4.368**	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637) -2.026 (1.242) 9.083***	Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717 (1.072) -0.418	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) -1.143*** (0.405) -1.977*** (0.724) 9.707***	10.49 e National Acco (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576) -2.524** (0.768) 10.128**	10.75 Dunting GPA of (Seme 10.75) -0. (0.1	Module 3 sester 1) 140 155) 9*** 339) 005 334) 18*** 333) 119 2236) 188 226) 118* 503) 9956 666) 53***	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502** (0.620) 9.841***	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239** (0.4621) 11.548***	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111 (0.731) 7.979***	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.303) -0.069 (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879 (0.593) 12.172***
F Variables / Grades Increase in literacy test scores Score to the first literacy test Age Gender (Men vs. women) Scholarship student Baccalaureate S Baccalaureate STG Other Baccalaureate Intercept	22.82 GPA of Module (Semester 1) 0.152 (0.174) 0.320*** (0.046) 0.029 (0.062) -0.848** (0.334) -0.547** (0.226) 1.577*** (0.260) -2.476*** (0.434) -1.861** (0.761) 7.118*** (1.443)	31.02 Mathematics (GPA) 0.395 (0.255) 0.310*** (0.070) 0.092 (0.086) -0.952* (0.504) -0.601* (0.327) 3.419*** (0.393) -3.292*** (0.604) -1.474 (1.057) 4.368** (2.056)	16.03 Mathematics (CA) -0.124 (0.278) 0.244*** (0.077) 0.037 (0.099) -1.736*** (0.564) -0.635* (0.359) 3.661*** (0.428) -3.560*** (0.637) -2.026 (1.242) 9.083*** (2.318)	16.1 Mathematics (FE) 0.836*** (0.298) 0.370*** (0.081) 0.168** (0.084) -0.190 (0.598) -0.320 (0.375) 3.287*** (0.449) -2.811*** (0.766) -0.717 (1.072) -0.418 (2.152)	English languag (GPA) 0.042 (0.165) 0.416*** (0.043) -0.028 (0.057) -0.303 (0.323) -0.499** (0.217) 0.319 (0.217) -1.143*** (0.405) -1.977*** (0.724) 9.707*** (1.265)	10.49 e National Accc (FE) -0.182 (0.244) 0.222** (0.063) -0.050 (0.076) -1.229** (0.447) -0.742* (0.310) -0.797* (0.351) -2.462** (0.576) -2.524** (0.768) 10.128** (1.836)	## 10.75 Dunting GPA of (Seme 10.22) ## 0.22 (0.1) ## -1.14 (0.1) ## -0.1 ## 0.0 ## 0.0 ## 10.8 (0.1) ## 10.8 (0.1) ## 10.8	Module 3 sester 1) 140	5.36 Methodology (GPA) 0.106 (0.145) 0.277*** (0.041) -0.052 (0.041) -0.763*** (0.277) -0.264 (0.185) -0.058 (0.212) -1.392*** (0.352) -1.502** (0.620) 9.841*** (0.932)	7.098 Methodology (CA) 0.072 (0.139) 0.278*** (0.038) -0.032 (0.054) -0.544* (0.286) -0.390** (0.190) -0.254 (0.201) -1.526*** (0.466) -1.239** (0.621) 11.548*** (1.128)	(FE) 0.027 (0.210) 0.318*** (0.057) -0.048 (0.041) -1.094*** (0.403) -0.156 (0.265) 0.219 (0.292) -0.783 (0.546) -1.111 (0.731) 7.979*** (1.137)	-0.146 (0.148) 0.268*** (0.040) -0.013 (0.034) -1.108*** (0.204) -0.163 (0.201) -1.784*** (0.649) -0.879 (0.593) 12.172*** (0.906)

	Second semester	GPA of Module 1	Introduction to	Introduction to	Introduction to	Introduction to	Introduction to	Introduction to	Financial	Financial	Financial
Variables / Grades	GPA	(Semester 2)	Microeconomics (GPA)	Microeconomics (CA)	Microeconomics (FE)	Macroeconomics (GPA)	Macroeconomics (CA)	Macroeconomics(FE)	economics (GPA)	Economics (CA)	economics (FE)
Increase in literacy test scores	0.287*	0.499*	0.997***	0.867**	0.769**	0.351	0.223	0.048	0.573*	0.640**	0.250
	(0.171)	(0.277)	(0.356)	(0.353)	(0.368)	(0.284)	(0.276)	(0.308)	(0.308)	(0.316)	(0.327)
Score to the first literacy test	0.265***	0.262***	0.323***	0.222**	0.318***	0.354***	0.209***	0.337***	0.284***	0.275***	0.243***
	(0.040)	(0.063)	(0.086)	(0.091)	(0.089)	(0.064)	(0.061)	(0.070)	(0.071)	(0.076)	(0.070)
Age	-0.007	0.000	0.103	0.043	0.113	0.068	-0.025	-0.012	0.088	0.028	-0.056
	(0.065)	(0.085)	(0.073)	(0.079)	(0.097)	(0.097)	(0.088)	(0.101)	(0.109)	(0.092)	(0.140)
Gender (Men vs. women)	-0.428	-0.523	-0.519	-0.379	-0.381	-0.565	-0.802	-0.034	-0.700	-0.541	-0.328
	(0.348)	(0.543)	(0.731)	(0.731)	(0.729)	(0.543)	(0.529)	(0.575)	(0.598)	(0.628)	(0.646)
Scholarship student	-0.387**	-0.364	-0.174	-0.163	-0.534	-0.045	-0.114	-0.509	-0.532	-0.632*	-0.866**
	(0.189)	(0.286)	(0.385)	(0.393)	(0.396)	(0.299)	(0.304)	(0.326)	(0.335)	(0.369)	(0.354)
Baccalaureate S	0.558**	0.595*	1.398***	2.202***	0.728	0.508	-0.041	1.114***	-0.481	-0.440	-0.455
	(0.223)	(0.348)	(0.467)	(0.447)	(0.473)	(0.372)	(0.372)	(0.380)	(0.388)	(0.404)	(0.397)
Baccalaureate STG	-1.818***	-2.471***	-1.808**	-1.369	-1.875**	-2.980***	-2.632***	-2.283***	-2.769***	-1.607**	-2.526***
	(0.441)	(0.607)	(0.760)	(0.877)	(0.872)	(0.630)	(0.637)	(0.673)	(0.651)	(0.761)	(0.769)
Other Baccalaureate	-0.394	-1.206	-0.111	0.372	0.361	-2.741***	-2.730**	-1.213	-2.808**	-1.767	-2.001
	(0.854)	(1.140)	(1.193)	(1.493)	(1.354)	(1.037)	(1.153)	(1.001)	(1.211)	(1.198)	(1.336)
Intercept	7.838***	6.218***	0.944	4.526*	0.964	4.707**	10.611***	5.548**	5.191**	7.688***	8.506***
*	(1.550)	(2.293)	(2.111)	(2.552)	(2.779)	(2.196)	(2.245)	(2.513)	(2.465)	(2.455)	(3.221)
Observations	443	466	493	454	451	492	453	453	488	454	452
R2	0.324	0.259	0.168	0.065	0.121	0.293	0.175	0.153	0.261	0.103	0.157
F	19.40	13.39	16.48	10.77	8.694	16.41	7.692	10.68	13.79	6.890	6.897

Variables / Grades	GPA of Module 2 (Semester 2)	Entrepreneurship (GPA)	Entrepreneurship (CA)	Entrepreneurship (FE)	Statistics and computer science (GPA)	Statistics and computer science (CA)	Statistics and computer science (FE)	GPA of Module 3 (Semester 2)	Principles Of law and right	English language (Semester 2)
Increase in literacy test scores	0.274	0.274	-0.054	-0.063	0.769***	0.402	0.607*	0.233	0.014	0.479**
	(0.216)	(0.251)	(0.202)	(0.268)	(0.297)	(0.248)	(0.330)	(0.165)	(0.220)	(0.231)
Score to the first literacy test	0.302***	0.243***	0.152***	0.198***	0.447***	0.298***	0.486***	0.217***	0.055	0.398***
	(0.049)	(0.056)	(0.047)	(0.063)	(0.070)	(0.056)	(0.076)	(0.036)	(0.045)	(0.052)
Age	0.024	0.125	0.050	-0.041	0.134	-0.017	0.038	-0.093***	-0.169***	-0.037
	(0.083)	(0.089)	(0.110)	(0.072)	(0.094)	(0.069)	(0.117)	(0.035)	(0.044)	(0.067)
Gender (Men vs. women)	-0.179	-0.487	-0.581	-0.579	0.461	0.552	0.294	-0.103	-0.215	0.246
	(0.420)	(0.465)	(0.408)	(0.529)	(0.590)	(0.477)	(0.628)	(0.322)	(0.403)	(0.438)
Scholarship student	-0.481**	-0.164	-0.718***	-0.396	-0.371	-0.560**	-0.701**	-0.092	0.050	-0.280
	(0.223)	(0.255)	(0.222)	(0.279)	(0.313)	(0.263)	(0.351)	(0.172)	(0.217)	(0.242)
Baccalaureate S	0.911***	-0.049	-0.113	0.115	1.271***	1.167***	1.593***	0.113	-0.186	0.232
	(0.280)	(0.324)	(0.253)	(0.345)	(0.406)	(0.329)	(0.433)	(0.206)	(0.263)	(0.287)
Baccalaureate STG	-1.739***	-2.172***	-1.243**	-1.247**	-2.548***	-1.783***	-2.268***	-0.967**	-1.163**	-0.811
	(0.453)	(0.619)	(0.506)	(0.502)	(0.657)	(0.563)	(0.717)	(0.463)	(0.566)	(0.669)
Other Baccalaureate	-1.735*	-3.079***	-2.444***	-0.607	-1.601	-1.587*	-0.259	-0.356	-0.348	-0.751
	(0.993)	(1.075)	(0.891)	(1.327)	(0.987)	(0.950)	(1.182)	(0.477)	(0.623)	(0.563)
Intercept	6.449***	5.631***	10.756***	9.394***	0.586	8.361***	1.168	10.899***	13.946***	8.027***
	(1.996)	(1.901)	(2.146)	(1.960)	(2.266)	(1.868)	(2.899)	(1.044)	(1.417)	(1.770)
Observations	469	491	462	454	489	439	455	447	454	449
R2	0.298	0.237	0.099	0.027	0.301	0.179	0.261	0.127	0.049	0.023
F	17.76	9.834	5.758	4.423	23.64	14.78	17.91	10.87	2.802	12.87

Source: randomized experiment implemented in University Paris-Est Marne-La-Vallée (Paris region, France; 2011-2014).

Field: 526 first-year university students in Economics and Management, entering the literacy learning device for the first year, for whom information from the baseline survey is available, as well as the two literacy test score.

Notes: effect of increasing (local average treatment effect; Wald estimator). For a considered teaching, FE stands for final exam. CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. Included baseline variables: score to the first literacy test, age of the university student; gender (reference=female); scholarship student; kind of baccalaureate (reference=Economics and Social Science stream). *** (respectively ** and *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: at a 5 percent level, a rise of 1 point in the difference between the second and the first literacy test scores induces an increase of 0.552 point in the final exam of Introduction to Economics and Management at university Paris -Est Marne-La-Vallée.

Table A7. Testing for differences in sample characteristics between tutorial groups that where encouraged to literacy learning and those who were not, for first-year university students in Economics and Management at university Lille 1.

Characteristics	Encouraged (1)	Not encouraged (2)	Difference (1)-(2) (significance)
Score to the first written literacy test ^a	6.1	6.1	0.0
Age ^b	19.6	19.4	0.2
Gender (being a man) ^c	58.9	67.3	-8.5
French nationality	78.2	82.4	-3.2
Scholarship student	51.6	50.7	-1.1
Kind of baccalauréat ^c :			
Bac ES (Economics and Social Science stream)	52.4	49.2	3.2
Bac S (Science stream)	29.0	32.7	-3.7
Bac STG (Technology stream)	9.7	9.5	0.2
Other (Bac L - Literature stream; foreign)	8.9	8.5	0.4
French department for the baccalauréat ^c :			
Nord	58.3	62.6	-4.3
Pas de Calais	10.8	9.1	1.7
Other French department	30.8	28.3	2.5
Abroad	17.5	18.2	-0.7
French department for residence ^c			
Nord	97.6	96.5	1.1
Pas de Calais	1.6	2.5	-0.9
Other (including abroad)	0.8	1.0	-0.2

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014).

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: a score; byears; cpercentage. *** (respectively ** or *) stands for significance of the difference at a 1% (respectively 5% or 10%) level.

Table A8. Effect of encouragement to literacy learning on academic performance of first-year university students at university Lille 1.

	Overall GPA	First semester	GPA of Module 1	Introduction	Introduction	Introduction	Methodology	GPA of Module 2	Introduction	Introduction to national	Defining caree
Variables / Disciplines	in first-year BA	GPA	(Semester 1)	to economics (GPA)	to economics (CA)	to economics(FE)	(Semester 1)	(Semester 1)	to management	accounting	objectives
Encouragement	0.407 (0.373)	0.564*	0.473 (0.330)	0.751* (0.388)	0.665 (0.432)	0.549 (0.522)	-0.393 (0.375)	0.944***	0.998* (0.529)	-0.304 (0.314)	1.606*** (0.310)
Intercept	9.156*** (0.246)	10.181*** (0.221)	10.594*** (0.218)	10.067*** (0.253)	9.890*** (0.266)	9.140*** (0.330)	11.947*** (0.233)	12.353*** (0.241)	11.422*** (0.358)	13.497*** (0.208)	13.799*** (0.227)
Observations	322	323	323	323	319	319	318	323	312	316	315
R2	0.004	0.008	0.006	0.011	0.007	0.003	0.003	0.021	0.011	0.003	0.069
F	1.194	2.908	2.056	3.750	2.370	1.108	1.103	7.787	3.552	0.939	26.80

	GPA of Module 3	Financials	Financial	Financial	Statistics	GPA of Module 4	Economic	Economic	GPA of Module 5	English	English	English
Variables / Disciplines	(Semester 1)	Mathematics(GPA)	Mathematics(CA)	Mathematics(FE)	(Semester 1)	(Semester 1)	sociology	History	(Semester 1)	(Semester 1, GPA)	(Semester, CA)	(Semester 1, FE)
Encouragement	0.156 (0.454)	0.040 (0.479)	0.602 (0.537)	-0.506 (0.509)	0.310 (0.479)	0.699 (0.505)	0.591 (0.550)	1.118* (0.575)	0.549 (0.404)	0.184 (0.504)	1.705* (0.990)	-0.624 (1.049)
Intercept	8.215***	8.117***	11.076***	4.973***	8.272***	9.776***	10.138***	9.856***	9.979***	9.619***	15.963***	20.074***
	(0.296)	(0.321)	(0.365)	(0.336)	(0.301)	(0.318)	(0.348)	(0.365)	(0.264)	(0.330)	(0.657)	(0.667)
Observations	323	322	315	312	322	323	310	312	323	322	284	308
R2	0.000	0.000	0.004	0.003	0.001	0.006	0.004	0.012	0.005	0.000	0.010	0.001
F	0.118	0.00702	1.259	0.986	0.418	1.916	1.153	3.775	1.842	0.133	2.964	0.354

	Second semester	GPA of Module 6	Introduction to	Methodology	GPA of Module 7	General	General	General	Functional analysis
Variables / Disciplines	GPA	(Semester 2)	Macroeconomics	(Semester 2)	(Semester 2)	accounting (GPA)	accounting (CA)	accounting (FE)	of organizations
-	0.076	0.215	0.407	0.160	0.122	0.100	0.406	0.150	0.120
Encouragement	0.276	0.315	0.487	-0.160	-0.123	-0.188	-0.496	-0.152	-0.139
	(0.449)	(0.551)	(0.596)	(0.397)	(0.520)	(0.448)	(0.462)	(0.460)	(0.644)
Intercept	8.104***	9.163***	10.027***	11.514***	8.294***	9.270***	11.782***	7.520***	9.232***
	(0.292)	(0.342)	(0.368)	(0.261)	(0.327)	(0.289)	(0.285)	(0.297)	(0.412)
Observations	322	322	269	297	322	301	286	292	277
R2	0.001	0.001	0.003	0.001	0.000	0.001	0.004	0.000	0.000
F	0.377	0.326	0.666	0.163	0.0559	0.177	1.154	0.110	0.0465

Variables / Disciplines	GPA of Module 8 (Semester 2)	Mathematics (GPA)	Mathematics (Exam 1)	Mathematics (Exam 2)	Mathematics (CA)	Statistics (Semester 2, GPA)	Statistics (FE)	GPA of Module 9 (Semester 2)	Big contemporary economic issues	Economic history	Tutorials in economics and computer science	GPA of Module 10 (Semester 2)	English (Semester 2, GPA)	English (Semester 2, CA)	English (Semester 2, FE)
Encouragement	0.283	0.087	-0.501	0.074	2.008***	0.133	-0.395	0.684	0.557	0.532	0.620	0.223	-0.414	-0.997	-0.909
	(0.496)	(0.500)	(0.526)	(0.485)	(0.732)	(0.596)	(0.619)	(0.482)	(0.497)	(0.638)	(0.425)	(0.496)	(0.519)	(1.054)	(0.942)
Intercept	5.672***	4.368***	4.298***	3.207***	5.875***	8.522***	7.177***	9.059***	9.095***	10.107***	10.054***	8.339***	8.971***	16.312***	16.742***
	(0.310)	(0.324)	(0.363)	(0.310)	(0.399)	(0.383)	(0.404)	(0.312)	(0.336)	(0.432)	(0.272)	(0.329)	(0.339)	(0.662)	(0.653)
Observations	322	277	284	274	248	292	288	322	278	276	317	322	293	251	273
R2	0.001	0.000	0.003	0.000	0.032	0.000	0.001	0.006	0.004	0.002	0.007	0.001	0.002	0.004	0.003
F	0.327	0.0304	0.906	0.0232	7.524	0.0494	0.408	2.009	1.258	0.696	2.130	0.202	0.634	0.894	0.932

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014) and Tables A9 to A11 (appendix).

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: at a 1 percent level, encouraging to literacy practice increases by 0.944 point the score to GPA of Module 2 for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table A9. Effect of increasing the literacy level on academic performance of first-year university students at university Lille 1. *Detailed results*.

Variables / Disciplines	Overall GPA In first year BA	First semester GPA	GPA of Module 1 (Semester 1)	Introduction to economics (GPA)	Introduction to economics (CA)	Introduction to economics(FE)	Methodology (Semester 1)	GPA of Module 2 (Semester 1)	Introduction to management	Introduction to National accounting	Defining Career objectives
		0.0454							. =	0.107	
Increase in literacy test scores	0.250 (0.213)	0.345*	0.290	0.460** (0.226)	0.414	0.341 (0.313)	-0.245 (0.246)	0.578***	0.568* (0.294)	-0.186	0.974***
Intercept	(0.213) 8.800***	(0.185) 9.694***	(0.191) 10.185***	(0.226) 9.419***	(0.258) 9.305***	(0.313) 8.657***	(0.246)	(0.201) 11.538***	(0.294) 10.648***	(0.203) 13.760***	(0.234) 12.426***
тистеері	(0.486)	(0.426)	(0.437)	(0.512)	(0.566)	(0.689)	(0.539)	(0.474)	(0.673)	(0.462)	(0.526)
Observations	322	323	323	323	319	319	318	323	312	316	315
R2	0.130	0.159	0.099	0.091	0.065	0.065	-0.112	0.088	0.079	-0.121	-0.300
F	1.377	3.444	2.283	4.098	2.547	1.181	0.982	8.214	3.719	0.830	17.20

Variables / Disciplines	GPA of Module 3 (Semester 1)	Financial Mathematics(GPA)	Financial Mathematics(CA)	Financial Mathematics(CT)	Statistics (Semester 1)	GPA of Module 4 (Semester 1)	Economic sociology	Economic History	GPA of Module 5 (Semester 1)	English (Semester 1.GPA)	English (Semester 1, CA)	English (Semester 1, FE)
v ariables / Disciplines	(Schiester 1)	Wathematics(GLA)	Wathematics(CA)	Wathematics(C1)	(Schiester 1)	(Schiester 1)	sociology	Ilistol y	(Schiester 1)	(Schester 1, Gr A)	(Semester 1, CA)	(Schester 1, 112)
Increase in literacy test scores	0.095	0.025	0.373	-0.306	0.189	0.428	0.344	0.674**	0.336	0.113	0.993*	-0.362
Intercept	(0.271) 8.080***	(0.290) 8.083***	(0.320) 10.545***	(0.331) 5.398***	(0.280) 8.007***		(0.305) 9.665***		(0.235) 9.505***	(0.303) 9.460***	(0.551) 14.505***	(0.623) 20.565***
	(0.611)	(0.664)	(0.744)	(0.740)	(0.616)	(0.645)	(0.682)	(0.733)	(0.528)	(0.684)	(1.295)	(1.386)
Observations	323	322	315	312	322	323	310	312	323	322	284	308
R2	0.042	0.009	0.067	-0.138	0.079	0.107	0.086	0.106	0.102	0.034	0.104	-0.056
F	0.123	0.00709	1.354	0.851	0.453	2.131	1.262	4.083	2.031	0.138	3.226	0.335

Variables / Disciplines	Second semester GPA	GPA of Module 6 (Semester 2)	Introduction to Macroeconomics	Methodology (Semester 2)	GPA of Module 7 (Semester 2)	General accounting (GPA)	General accounting (CA)	General accounting (FE)	Analyse fonctionnelle des organisations
variables / Disciplines	OFA	(Semester 2)	Macroeconomics	(Semester 2)	(Semester 2)	accounting (GFA)	accounting (CA)	(PE)	ues organisations
Increase in literacy test scores	0.170	0.193	0.288	-0.098	-0.075	-0.117	-0.308	-0.092	-0.082
	(0.264)	(0.327)	(0.340)	(0.248)	(0.324)	(0.286)	(0.310)	(0.282)	(0.382)
Intercept	7.864***	8.889***	9.569***	11.662***	8.401***	9.446***	12.262***	7.660***	9.363***
	(0.596)	(0.719)	(0.812)	(0.581)	(0.716)	(0.663)	(0.715)	(0.670)	(0.931)
Observations	322	322	269	297	322	301	286	292	277
R2	0.079	0.058	0.063	-0.036	-0.034	-0.064	-0.168	-0.044	-0.024
F	0.410	0.347	0.710	0.157	0.0540	0.166	0.983	0.105	0.0454

Variables / Disciplines	GPA of Module 8 (Semester 2)	Mathematics (GPA)	Mathematics (Exam 1)	Mathematics (Exam 2)	Mathematics (CA)	Statistics (Semester 2, GPA)	Statistics (Semester 2, FE)	GPA of Module 9 (Semester 2)	Grands problèmes économiques contemporains	Economic history	Tutorials in economics and computer science	GPA of Module 10 (Semester 2)	English (Semester, GPA)	English (Semester 2, CA)	English (Semester 2, FE)
Increase in literacy test scores	0.174	0.054	-0.309	0.043	1.419**	0.082	-0.244	0.420	0.348	0.328	0.391	0.137	-0.250	-0.600	-0.529
scores	(0.293)	(0.304)	(0.341)	(0.278)	(0.561)	(0.359)	(0.404)	(0.275)	(0.312)	(0.382)	(0.240)	(0.295)	(0.331)	(0.676)	(0.577)
Intercept	5.425*** (0.647)	4.283*** (0.732)	4.786*** (0.848)	3.140*** (0.675)	3.424*** (1.254)	8.399*** (0.835)	7.553*** (0.949)	8.463*** (0.619)	8.533*** (0.769)	9.585*** (0.942)	9.498*** (0.538)	8.145*** (0.677)	9.357*** (0.786)	17.299*** (1.633)	17.541*** (1.431)
Observations	322	277	284	274	248	292	288	322	278	276	317	322	293	251	273
R2	0.069	0.016	-0.110	0.013	-0.143	0.032	-0.107	0.136	-0.005	0.055	0.185	0.051	-0.106	-0.129	-0.109
F	0.351	0.0308	0.812	0.0235	6.355	0.0511	0.364	2.315	1.233	0.733	2.635	0.214	0.568	0.780	0.834

Source: randomized experiment implemented in University Lille I (Northern France; 2013-2014).
Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of varying literacy test score (local average treatment effect; Wald estimator). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: at a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0,460 point in the GPA in introduction in economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table A10. Effect of ancouragement to literacy learning on academic performance of first year university students at university I ille 1

Partiables Disciplines In first year BA GPA (Semester I) to economics (GPA to economics (CA) to economics (FEA	Defining
Variables Disciplines In first year BA GPA (Semester 1) to economics (GPA) to economics (CA) to economics (FE) (Semester 1) to management national accounting Caree	1.533*** (0.305) 0.188*** (0.056) -0.163 (0.127) -0.791** (0.315)
Encouragement 0.413 0.533* 0.428 0.747** 0.693* 0.574 -0.512 0.889*** 0.938* -0.283 I. (0.338) (0.290) (0.300) (0.359) (0.399) (0.491) (0.358) (0.323) (0.521) (0.291) (0.360) (0.68) (0.062) (0.064) (0.078) (0.090) (0.109) (0.109) (0.070) (0.065) (0.107) (0.060) (0.068) (0.062) (0.064) (0.078) (0.090) (0.109) (0.070) (0.065) (0.107) (0.060) (0.061) (0.149) (0.130) (0.138) (0.165) (0.177) (0.233) (0.139) (0.135) (0.198) (0.125) (0.149) (0.358) (0.313) (0.323) (0.323) (0.399) (0.414) (0.536) (0.369) (0.347) (0.543) (0.314) (0.358) (0.313) (0.323) (0.399) (0.414) (0.536) (0.369) (0.347) (0.543) (0.314) (0.358) (0.313) (0.323) (0.399) (0.414) (0.536) (0.369) (0.347) (0.543) (0.314) (0.358) (0.399) (0.347) (0.543) (0.314) (0.358) (0.399) (0.348) (0.355) (0.362) (0.362) (0.395) (0.487) (0.348) (0.327) (0.510) (0.510) (0.360) (0.362) (0.395) (0.487) (0.348) (0.327) (0.510) (0.510) (0.360) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.348) (0.355) (0.348) (0.348) (0.355) (0.488) (0.360) (0.362)	1.533*** (0.305) 0.188*** (0.056) -0.163 (0.127) -0.791** (0.315)
Score to the first literacy test 0,427*** 0,425** 0,366** 0,348** 0,374*** 0,374*** 0,370*** 0,297*** 0,372*** 0,372*** 0,320*** 0,0660 (0,066) (0,066	(0.305) 0.188*** (0.056) -0.163 (0.127) -0.791** (0.315)
(0.38) (0.290) (0.300) (0.359) (0.399) (0.491) (0.358) (0.323) (0.521) (0.291) (0.291) (0.20	0.188*** (0.056) -0.163 (0.127) -0.791** (0.315)
Score to the first literacy test	0.188*** (0.056) -0.163 (0.127) -0.791** (0.315)
Company Comp	-0.163 (0.127) -0.791** (0.315)
Age	-0.163 (0.127) -0.791** (0.315)
Gender (Man vs. woman)	(0.127) -0.791** (0.315)
Gender (Man vs. woman) -0.577 - 0.677** - 0.097 0.040 -0.131 0.290 -0.309 -0.890** - 1.132** - 0.647** - 0.0358) (0.313) (0.323) (0.390) (0.414) (0.536) (0.369) (0.347) (0.543) (0.543) (0.514) (0.543) (-0.791** (0.315)
Comparison of the comparison	(0.315)
Scholarship student	
Baccalaureate S	
Baccalaureate S	(0.310)
Colored Colo	-0.058
Baccalaureate STG -2.089*** -2.061*** -1.648*** -2.397*** -2.212*** -3.180*** -0.091 -1.789*** -2.634*** -1.287*** (0.557) (0.443) (0.553) (0.640) (0.671) (0.842) (0.590) (0.561) (0.922) (0.485) (0.645) (0.646) (0.671) (0.842) (0.590) (0.561) (0.922) (0.485) (0.646) (0.671) (0.842) (0.671) (0.842) (0.590) (0.561) (0.922) (0.485) (0.646) (0.748) (0.716) (0.657) (0.808) (0.931) (1.077) (0.660) (0.723) (1.088) (0.611) (0.672) (0.842) (0.842) (0.	(0.386)
Other baccalaureate (0.557) (0.443) (0.553) (0.640) (0.671) (0.842) (0.590) (0.561) (0.922) (0.485) (0.641) (0.716) (0.716) (0.800) (0.716) (0.808) (0.931) (1.077) (0.660) (0.723) (1.088) (0.611) (0.657) (0.808) (0.931) (1.077) (0.660) (0.723) (1.088) (0.611) (0.657) (0.808) (0.931) (1.077) (0.660) (0.723) (1.088) (0.611) (0.657) (0.808) (0.931) (1.077) (0.660) (0.723) (1.088) (0.611) (0.657) (0.660) (0.723) (1.088) (0.611) (0.657) (0.660) (0.723) (1.088) (0.611) (0.657) (0.660) (0.723) (1.088) (0.611) (0.657) (0.660) (0.723) (1.088) (0.611) (0.657) (0.660) (0.723) (1.088) (0.611) (0.660) (0.723) (0.660) (0.723) (1.088) (0.611) (0.660) (0.723) (1.088) (0.611) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (0.660) (0.723) (-0.797
Other baccalaureate	(0.496)
(0.748)	0.758
Intercept 10.460*** 10.588*** 14.314*** 13.890*** 13.796*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.018*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 14.139*** 11.250*** 10.979*** 16.165*** 12.018*** 10.979	(0.544)
(2.941) (2.569) (2.752) (3.319) (3.550) (4.663) (2.720) (2.649) (3.928) (2.434) (2.569	16.495***
R2 0.220 0.270 0.213 0.180 0.170 0.149 0.157 0.158 0.111 0.148 F 10.20 14.77 8.860 8.557 8.247 6.882 6.846 6.581 4.257 8.099 GPA of Module 3 Financial Financial Statistics GPA of Module 4 Economic GPA of Module 5 Anglais du Angl	(2.541)
R2 0.220 0.270 0.213 0.180 0.170 0.149 0.157 0.158 0.111 0.148 F 10.20 14.77 8.860 8.557 8.247 6.882 6.846 6.581 4.257 8.099 GPA of Module 3 Financial Financial Statistics GPA of Module 4 Economic GPA of Module 5 Anglais du Anglai	309
F 10.20 14.77 8.860 8.557 8.247 6.882 6.846 6.581 4.257 8.099 GPA of Module 3 Financial Financial Statistics GPA of Module 4 Economic GPA of Module 5 Anglais Anglais du An	0.142
GPA of Module 3 Financial Financial Statistics GPA of Module 4 Economic GPA of Module 5 Anglais Anglais du An	7.938
ę ę	7.250
Variables / Disciplines (Semester 1) Mathematics(GPA) Mathematics(CA) Mathematics(FE) (Semester 1) (Semester 1) sociology history (Semester 1) du S1(F) S1 (CC) S	Anglais du
	S1 (CT)
Encouragement 0.262 0.112 0.640 -0.317 0.412 0.636 0.476 1.225** 0.451 0.098 1.701* -	-0.680
	(0.882)
	1.794***
	(0.178)
	-0.225
•	(0.371)
	-0.708
	(0.880)
	-2.367***
	(0.862)
(0.486) (0.528) (0.603) (0.542) (0.502) (0.527) (0.527) (0.592) (0.422) (0.500) (1.082) (0.500)	-0.888

			~		CDA -CM 11 C	Total 2	Maria	CDA CM : :	. 7	C 1	C 1	December 1 1 1	-	
	Vorint	oles / Disciplines	Se	econd semester GPA	GPA of Module 6 (Semester 2)	Introduction to Macroeconomic		GPA of Module (Semester 7)		General PA) Accounting (CA)	General	Functional analysis of organizations		
	Variat	nes / Discipililes		GFA	(Selliester 2)	Macroeconomic	s (Semester 2)	(Semester /)	Accounting (Gr	A) Accounting (CA	Accounting (FE)	of organizations	-	
	Enco	uragement		0.314	0.394	0.642	-0.248	-0.026	-0.104	-0.287	-0.059	-0.052		
	Linco	aragement		(0.425)	(0.536)	(0.572)	(0.375)	(0.505)	(0.446)	(0.458)	(0.458)	(0.617)		
	Score	to the first liter	acy test	0.439***	0.364***	0.388***	0.277***	0.446***	0.302***	0.293***	0.318***	0.520***		
	beore	to the mot mer	acy test	(0.082)	(0.101)	(0.118)	(0.078)	(0.105)	(0.088)	(0.085)	(0.097)	(0.137)		
	Age			-0.194	-0.295	0.135	-0.208	-0.208	0.156	0.130	0.205	-0.095		
	7150			(0.182)	(0.227)	(0.233)	(0.157)	(0.230)	(0.172)	(0.177)	(0.183)	(0.300)		
	Gend	er (Man vs. woi	man)	-0.431	-0.004	0.598	-1.344***	0.296	-0.668	-0.552	-0.416	0.854		
	Gena	er (mair vs. wor	iluii)	(0.443)	(0.538)	(0.568)	(0.357)	(0.521)	(0.461)	(0.457)	(0.492)	(0.613)		
	Schol	arship student		-0.791*	-0.905*	-0.872	-0.910**	-0.463	-1.014**	-0.837*	-0.838*	-0.399		
	School	arship student		(0.422)	(0.533)	(0.568)	(0.378)	(0.492)	(0.430)	(0.442)	(0.444)	(0.607)		
	Racca	alaureate S		0.045	-0.661	-0.053	-0.433	-0.213	0.912*	1.417***	0.925*	-0.817		
	Dacca	naureate 5		(0.510)	(0.637)	(0.662)	(0.441)	(0.589)	(0.504)	(0.495)	(0.523)	(0.723)		
	Dagas	alaureate STG		-2.107***	-2.188**	-2.426**	-1.102	-1.773**	, ,	, ,	0.061	-3.113**		
	Басса	naureate STG							-0.381	-0.178				
	041	. h. a a a a l a suma - + -		(0.726)	(0.906)	(1.064)	(0.748)	(0.898)	(0.835)	(0.853)	(0.828) 2.238**	(1.287)		
	Other	baccalaureate		-0.010	-0.837	-0.861	0.035	0.212	1.375	0.961		-0.871		
	τ			(0.848)	(1.063)	(1.171)	(0.771)	(1.036)	(0.895)	(0.883)	(0.932)	(1.167)		
	Interc	ept		10.024***	13.578***	5.226	15.429***	9.828**	4.941	7.622**	1.757	8.023		
				(3.597)	(4.511)	(4.688)	(3.128)	(4.587)	(3.415)	(3.504)	(3.710)	(5.942)		
	Obser	rvations		315	315	265	291	315	294	280	285	271	-	
	R2	vations		0.160	0.097	0.088	0.134	0.103	0.101	0.103	0.102	0.116		
	F			6.609	4.091	3.225	5.506	3.706	3.973	5.167	4.010	3.317		
				0.009	4.091	3.223	3.300	3.700	3.913	3.107	4.010	3.317	-	
Variables / Disciplines	GPA of Module 8 (Semester 2)	Mathematics (GPA)	Mathematic (Exam 1)	Mathematics (Exam 2)	Mathematics (CA)	Statistics (Semester 2, (GPA)	Statistics Semester 2, FE)		Big contemporary economic issues	history econo	rials in GPA mics and Modul er science (Semes	e 10 (Semester 2,	English (Semester 2, CA)	English (Semester 2 FE)
Encouragement	0.372 (0.462)	0.402 (0.439)	-0.139 (0.442)	0.272	2.486*** (0.698)	0.367	-0.240 (0.587)	0.663	0.592 (0.483)		562 0.16 358) (0.46		-0.998	-0.903
C + - +1				(0.415)		(0.565)	(/	(0.451)					(1.037)	(0.844)
Score to the first literacy test	0.308***	0.342***	0.453***		0.102		0.340***	0.523***			50*** 0.557		0.728***	1.375***
	(0.099)	(0.101)	(0.099)	(0.099)	(0.148)	(0.115)	(0.121)	(0.086)	(0.106)		076) (0.08		(0.223)	(0.173)
Age	-0.246	-0.089	0.052	-0.085	-0.528*	-0.134	-0.075	-0.112	-0.126		091 -0.1		-0.111	0.287
	(0.182)	(0.174)	(0.172)	(0.170)	(0.290)	(0.223)	(0.220)	(0.175)	(0.188)		152) (0.20		(0.446)	(0.365)
Gender (Man vs. woman)	-1.322***	-1.130**	-1.115**	-1.449***	-0.646	-1.977***	-1.907***	-0.455	1.589***	-1.474** -0.	722* -0.6	73 -0.988*	-1.888*	0.044
	(0.480)	(0.471)	(0.481)	(0.465)	(0.716)	(0.566)	(0.591)	(0.466)	(0.503)	(0.622) (0.622)	370) (0.49	(0.505)	(1.045)	(0.876)
Scholarship	-1.020**	-1.504***	-1.848***	* -1.173***	-1.923***	-1.038*	-1.336**	-0.785*	-0.388	-0.946 -0	.129 -0.79	0* -1.116**	-1.017	-2.093**
student														
	(0.439)	(0.420)	(0.436)	(0.396)	(0.665)	(0.547)	(0.564)	(0.449)	(0.476)	(0.630) (0.	362) (0.47)	(1) (0.487)	(0.992)	(0.860)
Baccalaureate S	2.424***	3.160***	3.962***		2.829***	1.716***	2.067***	-0.593	-1.183**		160 -0.7		-1.001	-0.159
	(0.545)	(0.541)	(0.572)	(0.524)	(0.833)	(0.638)	(0.662)	(0.542)	(0.553)	(0.720) (0.720)	426) (0.56	(0.579)	(1.174)	(1.033)
Baccalaureate	-2.124***	-1.101**	-1.096**		-1.171		2.905***	-2.043**		, ,	.913 -2.409		-4.258**	-4.059***
STG	_ .		,0								2.102	, 0	50	,
	(0.569)	(0.483)	(0.514)	(0.485)	(0.978)	(0.853)	(0.863)	(0.794)	(0.921)	(1.103) (0.	717) (0.90	(0.856)	(1.850)	(1.180)
Other baccalaureate	1.621*	1.555*	1.905*	1.087	1.266	2.012*	2.433**	-0.611			34** -0.4		-0.016	0.728
oaccaiaurcaic	(0.931)	(0.934)	(0.991)	(0.723)	(1.266)	(1.150)	(1.193)	(0.811)	(0.855)	(1.226) (0.	727) (0.86	(0.949)	(2.043)	(1.833)
Intercept	9.184**	4.212	0.688	2.803	15.719***	10.222**	7.812*	9.123***	8.285**		114 8.424		16.416*	4.057
тистеері	(3.632)	(3.442)	(3.451)	(3.336)	(5.825)	(4.351)	(4.313)	(3.444)	(3.690)		990) (4.02		(8.947)	(7.198)
	(3.032)	(3.772)	(3.731)	(3.330)	(3.023)	(1.551)	(1.515)	(3.777)	(3.070)	(1.003)	770) (4.02	(4.051)	(0.771)	(7.170)
Observations	315	271	279	270	243	286	282	315	271	270	10 31:	5 286	245	266

R2	0.217	0.289	0.353	0.339	0.157	0.180	0.188	0.165	0.144	0.104	0.310	0.173	0.186	0.119	0.261
F	11.09	15.16	17.31	16.02	6.835	8.781	8.688	6.895	5.294	3.814	18.68	8.709	8.468	3.992	14.70

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014) and Tables A9 to A11 (appendix).

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of varying literacy test score (*local average treatment effect*; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student; the student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. **** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level. Reading: at a 5 percent level, encouraging to literacy practice increases by 0.747 point the score to GPA in introduction to economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table A11. Effect of increasing the literacy level on academic performance of first-year university students at university Lille 1. *Information provided by the administrative baseline survey are used are explanatory variables of the considered score.* Detailed results.

Variables / Disciplines	Overall GPA In first year BA	First semester GPA	GPA of Module 1 (Semester 1)	Introduction to economic(GPA)	Introduction to economics(CA)	Introduction to economics(FE)	Methodology (Semester 1)	GPA of Module 2 (Semester 1)	Introduction to Management	Introduction to National accounting	Defining Career objective
Increase in literacy test scores	0.270	0.347**	0.278	0.486**	0.454*	0.376	-0.337	0.578***	0.563*	-0.183	0.984***
	(0.201)	(0.170)	(0.181)	(0.220)	(0.248)	(0.306)	(0.254)	(0.207)	(0.307)	(0.196)	(0.247)
Score to the first literacy test	0.416***	0.409***	0.353***	0.325***	0.350***	0.330***	0.386***	0.269***	0.346***	0.329***	0.133*
•	(0.062)	(0.057)	(0.061)	(0.076)	(0.086)	(0.105)	(0.074)	(0.065)	(0.105)	(0.062)	(0.075)
Age	-0.184	-0.141	-0.294**	-0.321**	-0.339**	-0.457**	-0.128	-0.208	-0.141	0.076	-0.276*
	(0.131)	(0.115)	(0.127)	(0.154)	(0.158)	(0.219)	(0.157)	(0.136)	(0.202)	(0.134)	(0.161)
Gender (Men vs. women)	-0.191	-0.189	0.294	0.724	0.504	0.815	-0.787	-0.077	-0.278	-0.918**	0.592
	(0.449)	(0.384)	(0.398)	(0.483)	(0.506)	(0.665)	(0.544)	(0.441)	(0.716)	(0.455)	(0.532)
Scholarship student	-0.687**	-0.648**	-0.574**	-0.669**	-0.178	-1.287***	-0.388	0.065	-0.158	-0.008	0.011
-	(0.307)	(0.261)	(0.278)	(0.337)	(0.378)	(0.452)	(0.376)	(0.315)	(0.488)	(0.315)	(0.388)
Baccalaureate S	0.003	-0.122	-0.899***	-0.721*	-0.217	-1.058*	-1.271***	-0.353	-0.115	0.190	0.063
	(0.374)	(0.314)	(0.326)	(0.405)	(0.444)	(0.550)	(0.421)	(0.371)	(0.589)	(0.377)	(0.455)
Baccalaureate STG	-1.811***	-1.706***	-1.363**	-1.900***	-1.672**	-2.733***	-0.426	-1.198*	-2.015**	-1.471***	0.171
	(0.557)	(0.453)	(0.557)	(0.624)	(0.719)	(0.874)	(0.659)	(0.619)	(0.978)	(0.528)	(0.673)
Other Baccalaureate	-0.515	-0.933	-0.930	-0.694	-2.111**	-0.311	-1.628**	-0.460	-0.214	-0.141	0.307
	(0.678)	(0.656)	(0.631)	(0.813)	(0.873)	(1.030)	(0.800)	(0.770)	(0.984)	(0.701)	(0.817)
Intercept	10.467***	10.647***	14.361***	13.973***	13.822***	16.187***	13.882***	14.237***	11.831***	10.956***	16.505***
	(2.618)	(2.251)	(2.532)	(3.077)	(3.189)	(4.361)	(3.004)	(2.524)	(3.743)	(2.569)	(3.134)
Observations	315	316	316	316	312	312	311	316	306	310	309
R2	0.331	0.384	0.292	0.244	0.218	0.203	-0.008	0.178	0.147	0.050	-0.275
F	11.94	17.99	9.747	9.528	8.892	7.387	5.875	6.610	4.442	7.186	4.503

	GPA of Module 3	Financial	Financial	Financial	Statistics	GPA of Module 4	Economic	Economic	GPA of Module 5	English	English	English
Variables / Disciplines	(Semester 3)	Mathematics(GPA)	Mathematics(CA)	Mathematics(FE)	(Semester 1)	(Semester 1)	Sociology	history	(Semester 1)	(Semester 1, GPA)	(Semester 1, CA)	(Semester 1, FE)
Increase in literacy test scores	0.171	0.073	0.412	-0.205	0.268	0.414	0.300	0.780**	0.293	0.064	1.057*	-0.408
nerease in meracy test scores	(0.250)	(0.274)	(0.311)	(0.304)	(0.262)	(0.282)	(0.307)	(0.327)	(0.215)	(0.275)	(0.560)	(0.538)
core to the first literacy test	0.444***	0.430***	0.342***	0.470***	0.457***	0.409***	0.452***	0.387***	0.571***	0.790***	0.955***	1.822***
core to the mist incracy test	(0.082)	(0.090)	(0.105)	(0.099)	(0.083)	(0.090)	(0.099)	(0.107)	(0.073)	(0.088)	(0.192)	(0.185)
ge	-0.121	0.014	-0.281	0.137	-0.256	0.069	0.073	-0.037	-0.149	-0.267	-0.789**	-0.163
8-	(0.167)	(0.196)	(0.204)	(0.200)	(0.158)	(0.165)	(0.189)	(0.188)	(0.160)	(0.177)	(0.394)	(0.384)
Gender (Men vs. women)	-0.435	-0.466	-0.432	-0.724	-0.406	-0.408	-0.675	0.614	-0.320	-0.617	-0.145	-1.328
,	(0.581)	(0.642)	(0.747)	(0.716)	(0.596)	(0.604)	(0.696)	(0.727)	(0.472)	(0.576)	(1.253)	(1.245)
cholarship student	-1.015***	-1.394***	-1.129**	-1.206***	-0.637	-0.945**	-0.821*	-1.438***	-0.771**	-1.116***	-1.634*	-2.441***
•	(0.388)	(0.421)	(0.495)	(0.460)	(0.404)	(0.439)	(0.489)	(0.516)	(0.333)	(0.427)	(0.880)	(0.881)
accalaureate S	2.232***	2.421***	2.437***	2.873***	2.043***	-0.889*	-1.372**	0.235	-0.703*	-0.524	-0.769	-0.915
	(0.464)	(0.514)	(0.568)	(0.555)	(0.476)	(0.503)	(0.578)	(0.581)	(0.395)	(0.499)	(1.043)	(0.998)
accalaureate STG	-1.364**	-1.035	0.277	-1.871***	-1.693**	-3.500***	-3.094***	-3.473***	-1.119*	-1.696**	-0.101	-5.191***
	(0.599)	(0.653)	(0.826)	(0.600)	(0.668)	(0.697)	(0.793)	(0.864)	(0.634)	(0.786)	(1.884)	(1.502)
Other Baccalaureate	0.947	0.812	0.029	1.798	1.082	-2.714***	-2.480***	-2.390**	-1.520**	-0.504	-1.786	0.313
	(0.938)	(1.057)	(1.123)	(1.148)	(0.889)	(0.954)	(0.871)	(1.116)	(0.774)	(0.914)	(1.531)	(2.034)
ntercept	7.677**	5.333	13.916***	-0.176	10.017***	6.936**	7.356**	7.865**	10.024***	11.212***	25.003***	15.511**
	(3.243)	(3.834)	(4.009)	(4.048)	(3.025)	(3.274)	(3.677)	(3.742)	(3.214)	(3.574)	(7.495)	(7.745)
Observations	316	316	309	306	316	316	303	305	316	315	277	301
R2	0.302	0.244	0.199	0.176	0.318	0.264	0.215	0.251	0.333	0.314	0.236	0.304
F	14.65	11.30	8.265	13.86	13.98	11.60	8.425	11.14	15.87	21.91	9.499	22.89

V-1-11/D'1-11	Second semester GPA	GPA of Module 6	Introduction to	Methodology	GPA of Module 7	General	General	General	Functional analysis
Variables / Disciplines	GPA	(Semester 6)	Macroeconomics	(Semester 2)	(Semester 2)	accounting (GPA)	accounting (CA)	accounting (FE)	of organizations
Increase in literacy test scores	0.205	0.258	0.395	-0.164	-0.017	-0.067	-0.182	-0.037	-0.032
	(0.260)	(0.332)	(0.329)	(0.250)	(0.327)	(0.289)	(0.298)	(0.285)	(0.374)
Score to the first literacy test	0.430***	0.353***	0.371***	0.286***	0.446***	0.305***	0.305***	0.320***	0.522***
	(0.076)	(0.096)	(0.111)	(0.080)	(0.103)	(0.087)	(0.086)	(0.097)	(0.134)
Age	-0.219	-0.326	0.055	-0.181	-0.206	0.170	0.171	0.212	-0.090
	(0.167)	(0.210)	(0.222)	(0.161)	(0.225)	(0.182)	(0.195)	(0.190)	(0.298)
Gender (Men vs. women)	-0.138	0.365	1.238*	-1.593***	0.271	-0.771	-0.828	-0.471	0.804
	(0.578)	(0.709)	(0.748)	(0.551)	(0.737)	(0.699)	(0.701)	(0.724)	(0.901)
Scholarship student	-0.745*	-0.847*	-0.813	-0.946**	-0.467	-1.032**	-0.879*	-0.847*	-0.405
	(0.397)	(0.505)	(0.535)	(0.383)	(0.493)	(0.434)	(0.455)	(0.447)	(0.608)
Baccalaureate S	0.082	-0.615	-0.034	-0.473	-0.216	0.895*	1.386***	0.917*	-0.821
	(0.480)	(0.609)	(0.630)	(0.454)	(0.581)	(0.505)	(0.505)	(0.520)	(0.712)
Baccalaureate STG	-1.897**	-1.923**	-1.993*	-1.270	-1.790*	-0.455	-0.362	0.019	-3.145**
	(0.741)	(0.938)	(1.082)	(0.785)	(0.969)	(0.917)	(0.933)	(0.916)	(1.326)
Other Baccalaureate	-0.107	-0.959	-1.163	0.110	0.220	1.407	1.049	2.264**	-0.854
	(0.784)	(1.012)	(1.125)	(0.787)	(1.052)	(0.925)	(0.954)	(0.961)	(1.187)
Intercept	10.029***	13.585***	5.798	15.299***	9.828**	4.830	7.260**	1.693	8.011
	(3.322)	(4.227)	(4.339)	(3.175)	(4.538)	(3.400)	(3.613)	(3.655)	(5.857)
Observations	315	315	265	291	315	294	280	285	271
R2	0.239	0.163	0.164	0.090	0.096	0.071	0.025	0.088	0.107
F	7.299	4.426	3.587	5.227	3.681	3,839	4.693	3,939	3,285

Variables / Disciplines	GPA of Module 8 (Semester 2)	Mathematics (GPA)	Mathematics (Exam 1)	Mathematics (Exam 2)	Mathematics (CA)	Statistics (Semester 2, GPA))	Statistics (Semester 2, FE)	GPA of Module 9 (Semester 2)	Big contemporary economic issues	Economic History	Tutorials in economics and computer science	GPA of Module 10 (Semester 2)	English (Semester 2, GPA)	English (Semester 2, CA)	English (Semester 2, FE)
Increase in literacy test scores	0.243	0.253	-0.087	0.166	1.789***	0.231	-0.155	0.434	0.376	0.368	0.374*	0.110	-0.296	-0.607	-0.536
Score to the first	(0.284)	(0.264)	(0.278)	(0.243)	(0.577)	(0.336)	(0.387)	(0.270)	(0.300)	(0.381)	(0.210)	(0.295)	(0.325)	(0.655)	(0.521)
literacy test	0.297***	0.329***	0.458***	0.373***	0.004	0.336***	0.346***	0.504***	0.456***	0.311**	0.727***	0.552***	0.575***	0.782***	1.414***
Age	(0.092)	(0.095)	(0.099)	(0.093)	(0.150)	(0.110)	(0.122)	(0.079)	(0.105)	(0.121)	(0.067)	(0.085)	(0.102)	(0.236)	(0.182)
	-0.275*	-0.142	0.069	-0.116	-0.882***	-0.183	-0.046	-0.164	-0.192	0.012	0.040	-0.124	0.250	0.006	0.384
	(0.167)	(0.172)	(0.179)	(0.168)	(0.305)	(0.224)	(0.234)	(0.159)	(0.192)	(0.240)	(0.137)	(0.194)	(0.224)	(0.510)	(0.389)
Gender (Men vs. women)	-0.975	-0.742	-1.250*	-1.197*	2.009	-1.619**	-2.145**	0.165	2.214***	-0.896	-0.184	-0.517 (0.674)	-1.452* (0.756)	-2.793* (1.453)	-0.777
Scholarship student	-0.965**	-1.428***	-1.871***	-1.147***	-1.619**	-0.978*	-1.362**	-0.688*	-0.370	-0.904	-0.042	-0.766*	-1.173**	-1.054	-2.141**
	(0.416)	(0.413)	(0.449)	(0.385)	(0.776)	(0.525)	(0.574)	(0.414)	(0.469)	(0.609)	(0.324)	(0.457)	(0.515)	(1.035)	(0.904)
Baccalaureate S Baccalaureate STG	2.467***	3.233***	3.941***	3.499***	3.452***	1.772***	2.047***	-0.516	-1.159**	-0.095	0.224	-0.711	-0.147	-0.982	-0.143
	(0.516)	(0.518)	(0.562)	(0.495)	(1.013)	(0.612)	(0.663)	(0.497)	(0.543)	(0.694)	(0.378)	(0.545)	(0.599)	(1.226)	(1.055)
	-1.874***	-0.826	-1.179**	-0.774	0.983	-2.883***	-3.082***	-1.597**	-1.046	-2.741**	-0.501	-2.297**	-2.918***	-4.819**	-4.589***
Other Baccalaureate	(0.612)	(0.533)	(0.580)	(0.529)	(1.543)	(0.896)	(0.979)	(0.794)	(0.983)	(1.162)	(0.693)	(0.936)	(0.979)	(2.067)	(1.374)
	1.506*	1.450*	1.944*	1.003	0.622	1.889*	2.519**	-0.816	-1.027	-2.863**	1.310**	-0.485	0.333	0.518	1.086
Intercept	(0.845)	(0.864)	(1.029)	(0.719)	(1.672)	(1.055)	(1.265)	(0.757)	(0.878)	(1.213)	(0.615)	(0.830)	(1.122)	(2.275)	(2.146)
	9.191***	4.603	0.577	3.002	18.019***	10.598**	7.640*	9.134***	8.605**	8.791*	4.266*	8.427**	2.913	15.390	3.296
	(3.346)	(3.219)	(3.435)	(3.195)	(5.652)	(4.123)	(4.382)	(3.098)	(3.619)	(4.642)	(2.562)	(3.841)	(4.296)	(9.779)	(7.549)
Observations	315	271	279	270	243	286	282	315	271	270	310	315	286	245	266
R2	0.287	0.327	0.334	0.365	-0.152	0.242	0.137	0.277	0.153	0.144	0.442	0.207	0.086	0.019	0.170
F	11.94	15.21	17.11	16.56	4.395	9.457	8.172	8.055	5.432	3.980	24.13	9.162	7.528	3.511	12.78

Source: randomized experiment implemented in University Lille 1 (Northern France; 2013-2014) and Tables A9 to A11 (appendix).

Field: 323 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests..

Notes: effect of encouragement to literacy learning (intention to treat; OLS estimator) or of varying literacy test score (local average treatment effect; Wald estimator). Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). Robust standard errors within parentheses. For a considered discipline, FE stands for final exam, CA for continuous assessment, and GPA for grade point average, computed as the average of both the scores obtained to both FE and CA. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 0,486 point in the GPA in introduction to economics for first-year university students in Economics and Management stream at university Lille 1 (Northern France).

Table A12a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for

first-year university students in Economic and Management. *Male students*.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encor	uragement to literacy learning (Intent	tion To treat, ITT)
Without	0.070	0.131*	0.064
baseline variables	(0.326)	(0.064)	(0.335)
With	0.094	0.136**(a)	0.085
baseline variables	(0.152)	(0.046)	(0.172)
Model	Effect of varying li	teracy test scores (Local Average Tre	eatment Effect, LATE)
Without	0.051	0.090**	0.046
baseline variables	(0.309)	(0.012)	(0.330)
With	0.066	0.091***(b)	0.060
baseline variables	(0.113)	(0.008)	(0.128)

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 207 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increases by 13.6 percentage points the probability to achieve the first term of the first-year university for male students in Economics and Management stream at university Paris-Est Marne-La-Vallée (Paris region, France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a rise of 9.1 percentage points the probability to achieve first-year university for male students in Economics and Management stream at university Lille 1 (Northern from France)

Table A12b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. Female students.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encou	ragement to literacy learning (Inter	ntion To treat, ITT)
Without	-0.037	0.016	-0.111
baseline variables	(0.693)	(0.857)	(0.222)
With	-0.056	0.005	-0.134(a)
baseline variables	(0.528)	(0.947)	(0.127)
Model	Effect of varying lit	eracy test scores (Local Average T	reatment Effect, LATE)
Without	-0.026	0.006	-0.056**(b)
baseline variables	(0.543)	(0.909)	(0.021)
With	-0.047	-0.003	-0.070***
baseline variables	(0.226)	(0.950)	(<0.001)

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 116 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the second term of the first-year university for female students in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces a decrease of 5.6 percentage points the probability to achieve the second term of first-year university for female students in Economics and Management stream at university Lille 1 (Northern from France).

Table A13a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:*

students whose country of origin is France.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encour	ragement to literacy learning (Inten	tion To treat, ITT)
Without baseline variables	0.043 (0.542)	0.100 (0.150)	0.006 (0.930)
With baseline variables	0.026 (0.679)	0.069(a) (0.282)	-0.001 (0.989)
Model	Effect of varying lite	eracy test scores (Local Average Tr	eatment Effect, LATE)
Without baseline variables	0.023 (0.571)	0.057 (0.111)	-0.001 (0.974)
With baseline variables	0.013 (0.756)	0.043 <i>(b)</i> (0.271)	-0.008 (0.846)

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 214 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the first term of the firstyear university for students (whose country of origin is France) in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores let unchanged the probability to achieve first-year university for students (whose country of origin is France) in Economics and Management stream at university Lille 1 (Northern from France)

Table A13b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample: students whose country of origin is NOT France.*

Discipline	Academic	First	Second	
	Year	Semester	Semester	
Model	Effect of encou	ragement to literacy learning (Intention To treat, ITT)		
Without	0.027	0.091	0.009	
baseline variables	(0.784)	(0.341)	(0.922)	
With	0.063	0.126	0.032(a)	
baseline variables	(0.448)	(0.116)	(0.694)	
Model	Effect of varying lite	eracy test scores (Local Average Tr	reatment Effect, LATE)	
Without	0.013	0.065	-0.002(b)	
baseline variables	(0.857)	(0.218)	(0.972)	
With	0.050	0.081**	0.027	
baseline variables	(0.371)	(0.024)	(0.644)	

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 109 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the first term of the first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 5 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces an increase of 8.1 percentage points the probability to achieve the first term of first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France).

Table A14a. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:*

students who got a baccalaureate with merit, honors or distinction.

Discipline	Academic Year	First Semester	Second Semester	
Model		ragement to literacy learning (Inten		
Without baseline variables	0.023 (0.781)	-0.005 (0.945)	-0.039 (0.650)	
With baseline variables	0.035 (0.433)	0.035(a) (0.622)	-0.031 (0.713)	
Model	Effect of varying lit	eracy test scores (Local Average Tr	reatment Effect, LATE)	
Without baseline variables	0.009 (0.842)	-0.002 (0.957)	-0.025 (0.507)	
With baseline variables	0.013 (0.756)	0.019(<i>b</i>) (0.633)	-0.027 (0.485)	

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 136 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at any level, encouraging to literacy practice let unchanged the probability to achieve the first term of the first-year university for students (whose country of origin is France) in Economics and Management stream at university Lille 1 (Northern from France). (b) At any level, an increase of 1 point in the difference between the second and the first literacy test scores let unchanged the probability to achieve first-year university for students (whose country of origin is France) in Economics and Management stream at university Lille 1 (Northern from France)

Table A14b. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample: students who got a baccalaureate without merit, honors or distinction.*

Discipline	Academic Year	First Semester	Second Semester
Model	Effect of encour	agement to literacy learning (Intent	tion To treat, ITT)
Without baseline variables	0.025 (0.730)	0.147** (0.042)	0.017 (0.782)
With baseline variables	0.019 (0.770)	0.144**(<i>a</i>) (0.037)	0.016 (0.783)
Model	Effect of varying lite	racy test scores (Local Average Tre	eatment Effect, LATE)
Without baseline variables	0.016 (0.768)	0.099*** (0.004)	0.010 (0.830)
With baseline variables	0.012 (0.804)	0.100***(<i>b</i>) (0.005)	0.011 (0.808)

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 187 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 5 percent level, encouraging to literacy practice increase by 14.4 percentage points the probability to achieve the first term of the first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces an increase of 10 percentage points in the probability to achieve the first term of first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France).

Table A14c. Evaluation of the impact of the encouragement to literacy learning, or of increasing literacy level on the probability of achieving semester or academic year for first-year university students in Economic and Management. *Considered sample:*

students who got a baccalaureate with pass 50%-60%.

Discipline	Academic	First	Second
	Year	Semester	Semester
Model	Effect of encou	aragement to literacy learning (Intent	ion To treat, ITT)
Without	-0.014	0.138	0.003
baseline variables	(0.884)	(0.145)	(0.969)
With	-0.010	0.152*(a)	0.016
baseline variables	(0.911)	(0.095)	(0.849)
Model	Effect of varying li	teracy test scores (Local Average Tre	eatment Effect, LATE)
Without	-0.024	0.130***	-0.001
baseline variables	(0.841)	(0.001)	(0.999)
With	-0.013	0.131***(b)	0.015
baseline variables	(0.895)	(0.001)	(0.881)

Source: randomized experiment implemented in University Lille 1 (Northern from France; 2013-2014).

Field: 118 first-year university students in Economics and Management, who participate in the literacy learning device, for whom information from the baseline administrative survey is available, as well as scores for the two literacy tests.

Notes: effect of encouragement to literacy learning (intention to treat; probit estimator) or of varying literacy test score (local average treatment effect; probit instrumental variable estimator). Marginal effects. Included explanatory variables: score to the first literacy test; age of the student; the student is a man; scholarship student; kind of baccalaureate (reference= Economics and Social Science stream). P-value within parentheses. *** (respectively ** or *) stands for significance at 1% (respectively at 5% or 10%) level.

Reading: (a) at a 10 percent level, encouraging to literacy practice increases by 15.2 percentage points the probability to achieve the first term of the first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France). (b) At a 1 percent level, an increase of 1 point in the difference between the second and the first literacy test scores induces an increase of 13.1 percentage points the probability to achieve the first term of first-year university for students (whose country of origin is NOT France) in Economics and Management stream at university Lille 1 (Northern from France).